

## The extended gate problem : intermodal hub location with multiple actors

*Citation for published version (APA):* Bouchery, Y., Fransoo, J. C., & Slikker, M. (2015). *The extended gate problem : intermodal hub location with* multiple actors. (BETA publicatie : working papers; Vol. 481). Technische Universiteit Eindhoven.

Document status and date: Published: 01/01/2015

#### Document Version:

Publisher's PDF, also known as Version of Record (includes final page, issue and volume numbers)

#### Please check the document version of this publication:

• A submitted manuscript is the version of the article upon submission and before peer-review. There can be important differences between the submitted version and the official published version of record. People interested in the research are advised to contact the author for the final version of the publication, or visit the DOI to the publisher's website.

• The final author version and the galley proof are versions of the publication after peer review.

• The final published version features the final layout of the paper including the volume, issue and page numbers.

Link to publication

#### General rights

Copyright and moral rights for the publications made accessible in the public portal are retained by the authors and/or other copyright owners and it is a condition of accessing publications that users recognise and abide by the legal requirements associated with these rights.

- · Users may download and print one copy of any publication from the public portal for the purpose of private study or research.
- You may not further distribute the material or use it for any profit-making activity or commercial gain
  You may freely distribute the URL identifying the publication in the public portal.

If the publication is distributed under the terms of Article 25fa of the Dutch Copyright Act, indicated by the "Taverne" license above, please follow below link for the End User Agreement:

www.tue.nl/taverne

#### Take down policy

If you believe that this document breaches copyright please contact us at:

openaccess@tue.nl

providing details and we will investigate your claim.



## The extended gate problem: Intermodal hub location with multiple actors

Yann Bouchery Jan Fransoo Marco Slikker

Beta Working Paper series 481

BETA publicatie	WP 481 (working paper)
ISBN ISSN	
NUR	804
Eindhoven	July 2015

## The extended gate problem: Intermodal hub location with multiple actors

Yann BOUCHERY\* <sup>(1)</sup>, Jan FRANSOO <sup>(2)</sup>, Marco SLIKKER <sup>(2)</sup>

(1) Ecole de Management de Normandie, Axe Logistique Terre Mer Risque 30 Rue de Richelieu, 76087 Le Havre, France

(2) Eindhoven University of Technology, School of Industrial Engineering P.O. Box 513, 5600 MB Eindhoven, the Netherlands

July 29, 2015

**Abstract:** Hinterland transportation has become increasingly critical for global container supply chain performance. However, the literature on the management of the container supply chain tends to overlook the issues related to hinterland transportation. The problem we consider here is faced by a deep-sea terminal operator who wants to open an inland terminal to facilitate its operations. When developed by a deep-sea terminal operator, such an inland terminal directly related to a deep-sea port is called an extended gate. In this setting, the shippers can decide to take their containers either directly at the deep-sea terminal or at the inland terminal. In this latter case, train transportation is used from the deep-sea terminal to the inland terminal. Intermodal hub location problems are usually solved by considering a single decision maker even though several actors often interact in practice. We analyze here the impact of having multiple actors involved by proposing a formulation of the problem based on game theory. We develop structural properties of the shippers' behavior. These properties enable us to identify the existing equilibria for the game and to solve the problem optimally. We apply the results to an example based on the features of the hinterland network in the Netherlands and we provide related insights. We show that the multiple actors feature of intermodal hinterland networks is critical and needs to be accounted for. Our results serve as a basis for appropriately taking multiple actors into account in hub location problems.

**Keywords:** hub location; intermodal transportation; hinterland supply chain; multiple actors; extended gate.

### **1. Introduction**

The vast majority of intercontinental freight transport takes place by sea and the amount of containerized goods is increasing. Containerization has indeed been the main technological revolution of the maritime industry in the past 30 years. We refer to Levinson (2010) for an analysis of the impact of containerization. This innovation has shaped current global supply chains by substantially reducing transportation costs. For example, the freight rate on a port-to-port basis between Shanghai and Rotterdam for a 40-foot container is 0.21 per km (OECD/ITF, 2009). This freight rate implies that the maritime transportation cost for 32-inch television screens from Asia to Europe is less than 3 per screen. As a result, global container traffic has been growing at almost three times world gross domestic product growth since the early 1990s (UN-ESCAP, 2005). The management of the container supply chain is consequently a key issue.

This article analyzes the efficiency of container transportation systems in the hinterland. Although the distance covered by the container in the hinterland is typically small, inland transportation costs are often substantial. For example, the freight rate for inland transportation by truck from the port of Rotterdam typically ranges from 1.50 per km to  $\oiint{4}$  per km, depending on the distance and weight (OECD/ITF, 2009); this is 7–19 times higher than the maritime transportation freight rate. Hinterland activities also include container handling operations. Summing up all the costs related to hinterland operations, Notteboom and Rodrigue (2005) estimate that the proportion of inland costs relative to the total transportation cost of a container shipping ranges from 40% to 80%. Thus, improving the efficiency of the hinterland supply chain could provide substantial benefits from a global supply chain perspective. Notteboom (2008a) accordingly mentions that the maritime industry have identified inland logistics as one of the most vital areas still left to cut costs.

When focusing on the hinterland supply chain, the main impact of containerization is the increasing role of intermodal transportation. Intermodal transportation involves the transportation of the load from origin to destination in the same transportation unit without handling of the goods themselves when changing modes (Crainic and Kim, 2007). The development of container based intermodal transportation leads to the development of inland terminals. This phase in the evolution of the hinterland supply chain is called "terminalization" and this is considered as a long-term trend (Rodrigue and Notteboom, 2009). The development of inland terminals is supported by several types of actors such as terminal operators, port authorities, rail operators, real estate promoters and/or local authorities. In addition to the actors directly involved in the development of inland terminals, the users of the designed service are of major importance for successful implementation. Indeed, the emergence of inland terminals can be seen as a compromise for which both operators and users find a valuable alternative to their constraints (Rodrigue and Notteboom, 2009). As a consequence, the performance of the hinterland network depends on the behavior of the different actors involved (De Langen and Chouly, 2004). This issue has been frequently emphasized and discussed in the maritime economics

and transport geography literature (Notteboom, 2008b; Roso et al., 2009; Van Der Horst and De Langen, 2008). However, model-based research on hinterland network design with multiple actors is scarce. Accordingly, Fransoo and Lee (2013) recently argued that research from the operations management and transportation science community is lacking on container transportation systems despite its critical role in current global supply chains.

In this article, we consider the problem faced by a deep-sea terminal operator who wants to facilitate its operations by creating a direct rail connection between the deep-sea terminal and an inland terminal where containers can be sorted by destination. This solution may be implemented in areas where the possibilities of expansion inside the port are limited or too costly and/or where road terminal access is an issue due to congestion. This strategy enables moving a share of terminal operations more inland where land availability and road terminal access are less of an issue. In addition, this solution favors intermodal transportation as a large volume of containers can be transported to the same location (Zuidwijk and Veenstra, 2014). When the project is implemented by a terminal operator, such an inland terminal directly connected to a deep-sea port is called an extended gate (Veenstra et al., 2012). The concept is currently developed in Europe but this one may also be of interest in other regions. We refer to Rodrigue and Notteboom (2009) for a description of an extended gate operated in the region of Venlo in the Netherlands. We summarize the decision process as follows. The terminal operator first decides on where to locate the extended gate. Then the shippers decide if they want to take their containers at the deep-sea terminal or at the extended gate. In case they decide to use the extended gate, they pay for train transportation and extra handling costs. In this leader-follower setting, the terminal operator needs to take the shippers' reactions into account when deciding on where to locate the extended gate.

We consider two objectives for the location decision made by the terminal operator. Besides minimizing the total cost as classically done in the literature, we argue that maximizing extended gate utilization is of great importance for the terminal operator. We show that these two objectives may be conflicting and we apply multiobjective optimization to identify the set of efficient solutions (also called Pareto optimal solutions). The shippers are primarily focused on minimizing transportation and handling costs incurred from hinterland deliveries in the setting we consider. The model takes flow dependent economies of scale into account for intermodal transportation. Thus, the decision of each shipper depends on the other shippers' decisions. We formulate the shippers' route choice decisions as a non-cooperative game. We identify key structural properties that enable identifying the existing equilibria. We highlight that several equilibria may exist and that one of these equilibria often involves not having any shipper using the extended gate. Thus, the terminal operator often needs to convince some shippers to use the extended gate in order to obtain the base volume necessary for becoming competitive. We propose a way to quantify this effort and we show that the optimal extended gate location may depend on the effort the terminal operator is willing to make. Finally, we identify the system optimum solution (i.e., the solution that would be obtained under full

centralization of decisions) in order to estimate the value of collaboration (as expressed by the socalled price of stability). We highlight that collaboration may be critical for successfully implementing an extended gate even if the value of collaboration may also be quite low. Overall, the results show that the multiple actors feature of intermodal hinterland networks is critical and needs to be accounted for.

We organize the rest of the article as follows: Section 2 analyzes the existing literature. Section 3 focuses on the description and the mathematical formulation of the model. Section 4 analyzes the structural properties of the problem. These properties enable us to identify the equilibria of the problem. Section 5 details the most important insights of our research by focusing on an example based on the features of the hinterland supply chain in the Netherlands. Finally, Section 6 offers conclusions.

### 2. Literature review

The maritime economics and transport geography literature related to the problem considered is presented in the introduction of the article. Therefore, we focus here on the model-based literature that serves as a basis for the development of the model. The problem we consider can be interpreted as a particular type of hub location problem. The first articles focusing on hub location can be traced back to 1986 (O'Kelly, 1986a, 1986b). The literature on hub location has since expanded rapidly. Alumur and Kara (2008), Campbell and O'Kelly (2012), and Farahani et al. (2013) all provide reviews. Several classical formulations of the hub location problem appear in the literature (e.g., hub median, hub center, hub covering), but the problem we consider here is related to the p-hub median problem. This problem consists of locating a given number of hubs and deciding how to allocate a set of origin/destination nodes to these hubs to minimize the total transportation cost of the system. The cost of transporting one unit of flow per unit distance is discounted on interhub arcs to represent the economies of scale achieved by such consolidation systems. This feature creates an incentive to route origin/destination flow through more than one hub because, though this increases the total distance traveled, it may lead to an overall cost reduction. The basic p-hub median model (and many of its extensions) assumes that economies of scale are somehow exogenous to the decisions made about hub location and origin/destination allocation. A fixed discount factor is typically applied to account for economies of scale on interhub arcs. This limitation was first addressed by O'Kelly and Bryan (1998), who account for flow-dependent economies of scale on interhub arcs by considering strictly increasing concave transportation cost functions. They prove that the optimal hub locations may differ greatly from the results obtained without taking flow-dependent economies of scale into account.

The p-hub median problem has been applied to study intermodal transportation networks. Bontekoning et al. (2004) present a review of the early research on intermodal transportation for hinterland supply chains. The literature on intermodal hub location has quickly expanded in the past decade (Alumur, Kara and Karasan, 2012; Alumur, Yaman and Kara, 2012; Arnold et al., 2004; Groothedde et al., 2005; Ishfaq and Sox, 2010, 2011, 2012; Jeong et al., 2007; Limbourg and Jourquin, 2009; Meng and Wang, 2011; Racunica and Wynter, 2005; Sörensen and Vanovermeire, 2013; Zhang et al., 2013). These articles primarily extend the classical p-hub median problem by taking classical features of intermodal container transportation into account. The most commonly considered aspects are flow-dependent economies of scale for transportation and transshipment activities, travel time, service constraints, and congestion in the system. These studies usually propose a new model that incorporates some features proven to be of practical importance. Then, they develop new solution techniques to solve the proposed problem and focus on assessing the efficiency of the considered solution technique.

To our knowledge, four articles consider multiple actors in intermodal hub location problems. Sirikijpanichkul et al. (2007) apply an agent-based modeling approach to optimize the location of intermodal freight hubs. They consider four types of agents, i.e., hub owners, transport network infrastructure provider, hub users and communities. Sörensen and Vanovermeire (2013) argue that, in general, the location and transportation costs typically included in intermodal hub location problems are not incurred by the same actors. Thus, they consider these two types of cost separately and develop a bi-objective optimization model to identify the existing trade-offs between both types of cost. Meng and Wang (2011) include user equilibrium constraints in a hub location problem to model the effects of having multiple stakeholders involved. They consider that a network planner designs the network with the objective of minimizing total costs and that the intermodal operators make their route choice decisions in order to minimize their own costs. The user equilibrium constraints, formulated as variational inequalities, ensure that no intermodal operator can reduce its transportation and transshipment cost by individually changing its routing strategy. Yamada et al. (2009) use bi-level programing and user equilibrium constraints to account for multiple actors in an intermodal network design problem. Bi-level programing provides results similar to a Stackelberg game in which the network planner first design the network and then the shippers make their routing decisions. The lower-level problem that focuses on routing decisions involves user equilibrium constraints as in Meng and Wang (2011) in order to account for multiple shippers.

These articles are particularly noteworthy here because they include a multiple-actors feature, which is often acknowledged as the most challenging aspect of container supply chains. However, they primarily focus on identifying effective procedures to solve the proposed models. Thus, the impact of having multiple actors involved in such supply chains cannot be easily assessed from their results. We attempt here to better understand the consequences of having multiple actors involved in hinterland supply chains. Our model is based on two main features. First, we consider that the hub location and the intermodal route choice decisions are not made by the same actors. We formulate a Stackelberg game in which the terminal operator first decides on where to open an extended gate, and shippers then allocate their demand flow in the network so as to minimize their individual

transportation and handling cost. Second, we consider that the shippers do not cooperate in making their route choices. Instead of using user equilibrium constraints as in Yamada et al. (2009) and Meng and Wang (2011), we formulate the shippers' routing decisions problem as a non-cooperative game. This game may be viewed as a special type of atomic congestion game as firstly introduced by Rosenthal (1973). We show that several equilibria may exist for the shippers' routing decisions game. We analyze the consequences of having multiple equilibria and we show that this feature is of primary importance. Yamada et al. (2009) is the first article using traffic assignment techniques in intermodal network design problem. We propose to follow up and to go further by modeling the shippers' routing decisions problem as a non-cooperative game. We derive key structural properties of the game and we identify all the existing equilibria. Our article is, to our knowledge, the first to apply directly non-cooperative game theory to analyze routing decisions in a hub location problems.

Our main contributions are fourfold. First, this article is, to our knowledge, the first to model the extended gate location problem. The extended gate location problem possesses some particular aspects not covered by the models developed in the existing literature. For instance, we argue in the next section that maximizing the extended gate utilization is of great importance for the terminal operator. The concept of extended gate is also highly relevant in practice. Second, we formulate the routing decisions made by the shippers as a non-cooperative game and we derive key structural properties of the game. This allows us to identify all the existing equilibria for the game. Third, we study the impact of having multiple actors involved in the extended gate location problem. We show how to compute the minimum number of shippers who need to be convinced to reach a given equilibria and we measure the value of collaboration by computing the price of stability. Fourth, we apply the results to an example based on the features of the hinterland network in the Netherlands and provide related insights.

### 3. Model description

### 3.1 Context

The model we propose describes the extended gate location and shippers' routing decisions problem. For the sake of clarity, we present the model in terms of import flows for containerized cargo. The containers are unloaded to a deep-sea terminal and need to be delivered to various destinations. The problem could be reversed by considering export flows from various origins to the deep-sea terminal. Our results hold in that case as well. Because the dimensions of containers have been standardized (Agarwal and Ergun, 2008), the proposed model takes only one type of container into account. We consider that the containers must be delivered to a fixed set of destinations with constant and known demand as this is usually the case in the hub location literature. This deterministic demand assumption comes from the fact that the unplanned share of the demand requires urgent delivery. Consequently,

the only feasible option in this case is to deliver by truck from the port. The problem under consideration does not take this share of the demand into account as this does not influence the decisions made and the results obtained. When arriving at their destination, the containers are unloaded. We do not explicitly take into account empty container management but consider this a parameter of the model.

The destinations are considered individual companies, such as retailers, requesting a transportation service in the model (merchant haulage setting). We will consequently refer to the shippers as the destinations in what follows. We consider that the destinations choose among two possible options for being delivered. Either they decide to take their containers at the port and have them delivered by truck, or they decide to use the extended gate. In that case, we assume that the containers are loaded on a train from the port to the extended gate. We focus on train transportation because this is the most developed intermodal transportation solution worldwide and because most of the current examples of extended gates are connected to the deep-sea terminal by rail. However, the model is also valid for barge and short sea intermodal transportation systems. After arriving at the extended gate, the containers are transshipped to trucks to reach their final destination. In case this latter option is chosen, the destinations are charged for the train transportation cost from the deep-sea terminal to the extended gate as well as for the extra handling costs. They also incur truck transportation cost for the final leg of the containers' journey.

We refer to the routing decisions made by the destinations as the shippers' allocation problem, by analogy with the hub location literature. This problem consists of deciding, for each destination, which share of the total flow should be allocated to the extended gate, i.e., shipped by intermodal transportation. In the literature on intermodal transportation, the decision of using intermodal transportation instead of direct truck delivery is mainly based on two attributes, the total cost and the leadtime. Indeed, intermodal transportation is traditionally perceived as less flexible and consequently implies longer leadtime compared to truck. In the setting considered here, we argue that the key attribute for explaining the destinations' decisions is the total cost for two main reasons. First, the increase in leadtime when using intermodal transportation compared to direct truck delivery is mainly due to the waiting time for train transportation at the port. However, the average dwell times at the main European ports are equivalent for vessel/truck and vessel/train transshipment (Rodrigue and Notteboom, 2009). This implies that the difference in leadtime is small. This small difference in leadtime would fall below the threshold of indifference from a shipping time perspective which is often evaluated to 3 days (Rodrigue and Guan, 2009). We recall here that we exclude urgent deliveries from the model. Second, the high dwell times at the port got increasingly associated with deliberate decisions of the destinations who use the terminals as low cost locations for keeping inventory. When using the extended gate, the containers can be directly shipped and stored at the extended gate and the remaining leadtime can even be lower than when using direct truck delivery from the port. Consequently, we assume here that the destinations decide on whether or not to use the extended gate by evaluating only the costs incurred for the two options. This feature has been validated with industrials.

Our model takes into account flow-dependent economies of scale for train transportation and transshipment operations at the extended gate. We model the train transportation cost and the transshipment cost as general non-negative function, non-increasing in the total number of containers shipped. As soon as these functions are non-constant, a cost interdependency among the destinations occurs, leading to some noteworthy allocation issues. In the classical single-actor vision of hub location problems, O'Kelly and Bryan (1998) point out that "some origin-destination pairs may be routed via a path that is not their least-cost path because doing so will minimize total network travel cost" (p. 608). This statement does not hold in a multiple-actor setting. The situation is similar to a classical problem in the traffic assignment literature. Because of congestion, the solution that minimizes the total travel time in the system is not equivalent to the solution that minimizes the travel times of each individual users. This leads to two extreme behaviors that Wardrop (1952) describes as user equilibrium (each user minimizes its own travel time) versus system optimum (the total travel time of the system is minimized). In our model, the destinations seek to minimize their own transportation and handling costs. Therefore, we refer to an allocation of the destinations' demand flow as a user equilibrium (UE) if no destination can decrease its transportation and handling cost by individually changing its routing decision and if intermodal transportation is used by a destination only if this strictly reduces its cost. In addition, we consider a system optimum (SO) allocation (i.e., an allocation that minimizes to total transportation and handling cost) in order to assess the value of collaboration.

The terminal operator decides where to locate an extended gate given a set of predetermined sites. The problem can be formulated as a p-hub median problem with a single origin and a single hub to locate (the extended gate). In the p-hub median problem, the objective is to minimize the total cost of the system. Our model assumes that the terminal operator and the different destinations are autonomous actors who make their decisions separately. By direct transposition, we could consider that the terminal operator aims at minimizing the total cost given the reactions of the destinations as in Yamada et al. (2009) and Meng and Wang (2011). Considering the total cost as the location objective makes sense as we may expect the extended gate to offer a competitive service if the total cost are minimized. On the other hand, the terminal operators also aim at maximizing the utilization of the designed service, given the reaction of the destinations. Indeed, such projects are very capital intensive and may be beneficial only if well utilized. In addition, one of the key motivation for the terminal operator to start such project is to transfer a share of terminal operations outside the port in order to increase its throughput. Indeed, the main customers of such deep-sea terminals are the shipping lines who are mainly focused on reducing the waiting time of their ships at the port. Moreover, shipping lines are charged by the terminal per container movement (loading or unloading). By increasing its throughput, the terminal operator can attract more traffic and increase its revenue.

Consequently, from a terminal operator's point of view, maximizing the utilization of the extended gate may be at least as important as minimizing the total cost of delivering the containers. In what follows, we consequently assess if the decision made by considering a total cost minimization objective are similar to the one taken when maximizing the extended gate utilization. Thus, we take this two objectives into account.

### **3.2 Model formulations**

All the notations used in the articles are summarized in Appendix A (available online as supplementary material). The hinterland supply chain under study consists of a single deep-sea terminal inside a port (considered as the origin) with  $N \in \mathbb{N}^*$  destinations ( $\mathbb{N}^*$  refers to the set of natural numbers). We consider a single period in the model as classical done in the hub location literature. A deterministic constant flow  $n_j$  must be shipped from origin to destination  $j \in \{1,...,N\}$ . Here,  $n_j$  is expressed in number of containers, and only one type of container is available (40-foot containers); thus, we assume that  $n_j \in \mathbb{N}^*$  for all  $j \in \{1,...,N\}$ . At most one extended gate must be located among  $M \in \mathbb{N}^*$  candidate locations. Each potential extended gate location is referred to as location  $i \in \{1,...,M\}$ . In addition, we refer to the origin (i.e., the deep-sea terminal) as location i = 0.

The truck transportation cost function considered is linear in the number of containers shipped and the cost of transporting one container from location  $i \in \{0,...,M\}$  to destination  $j \in \{1,...,N\}$  by truck is referred to as  $Z_{i,j}^1$ . This cost depends on the distance from *i* to *j*, on the rate proposed by the truck operators, on the empty container management practice developed by the destination and on the level of congestion around location *i*.

While delivery takes place through intermodal transportation, train transportation is used from the port to the extended gate. We define the cost of shipping one container by train from the port to the extended gate  $i \in \{1,...,M\}$  as  $Z_{0,i}^2(K)$ , where  $K \in \mathbb{R}^*_+$  is the total number of containers shipped by train ( $\mathbb{R}^*_+$  is the set of strictly positive real numbers). Flow-dependent economies of scale are taken into account; consequently, we assume that  $Z_{0,i}^2(K)$  is non-negative for all  $K \in \mathbb{R}^*_+$  and non-increasing in K, for all  $i \in \{1,...,M\}$ . Empty container are usually stored in empty depots outside the deep-sea terminal. Thus, empty containers cannot be shipped back to the empty depot by using the same train. In addition, if an export match is found, the containers are usually shipped back by truck due to time consideration. We can consequently assume train is used solely for shipping container from the deep-sea terminal to the extended gate.

In addition to train and truck transportation costs, intermodal transportation implies additional container handling operations at the extended gate. We define the cost of transshipping one container at extended gate  $i \in \{1,...,M\}$  as  $Z_i^{1,2}(K)$ . Note that the total number of containers transshipped at

extended gate *i* is equal to the total number of containers shipped by train. The transshipment cost depends on the location considered to account for the difference in land and labor costs as well as for the difference in extended gate layout, equipment, and size. This cost also accounts for the difference in cost from transshipping to a truck or to a train at the origin, if any. In addition, we consider that  $Z_0^{1,2}(K) = 0$ , for all  $K \in \mathbb{R}^*_+$ . The transshipment cost per container depends on the total amount of containers transshipped at extended gate *i*. We further assume that  $Z_i^{1,2}(K)$  is non-negative for all  $K \in \mathbb{R}^*_+$  and non-increasing in *K*, for all  $i \in \{1,...,M\}$ .

The total cost of shipping one container from the port to destination  $j \in \{1,...,N\}$  by intermodal transportation through extended gate  $i \in \{1,...,M\}$  is equal to  $Z_{0,i}^2(K) + Z_i^{1,2}(K) + Z_{i,j}^1$ . To simplify the notation, we define  $Z_{0,i}^3(K)$  as follow:

$$Z_{0,i}^{3}(K) = Z_{0,i}^{2}(K) + Z_{i}^{1,2}(K) \text{, for all } i \in \{0, \dots, M\}, \text{ for all } K \in \mathbb{R}_{+}^{*}.$$
(1)

Note that  $Z_{0,0}^3(K) = 0$  for all  $K \in \mathbb{R}^*_+$ . By applying the properties defined for  $Z_{0,i}^2(K)$  and  $Z_i^{1,2}(K)$ , we can state that  $Z_{0,i}^3(K)$  is non-negative for all  $K \in \mathbb{R}^*_+$  and non-increasing in K, for all  $i \in \{1,...,M\}$ . As is usually the case in the hub location literature, we further define  $0 \le X_{i,j} \le 1$  as the proportion of flow from the port to destination j being routed through extended gate i. Here,  $X_{0,j} = 1$ indicates that the entire flow from the port to destination j is delivered by direct shipment, while  $X_{i,j} = 1$ , where  $i \in \{1,...,M\}$ , indicates that the entire flow from the port to destination j is delivered by intermodal transportation using extended gate i.

The first objective considered consists in minimizing the total transportation and transshipment cost. This objective function may be expressed as follows:

$$MIN\sum_{i=0}^{M}\sum_{j=1}^{N}n_{j}\left(Z_{0,i}^{3}(K)+Z_{i,j}^{1}\right)X_{i,j},$$
(2)

The following set of constraints must be considered:

$$\sum_{i=1}^{M} y_i \le 1,\tag{3}$$

$$y_i \in \{0;1\}, \forall i \in \{1,...,M\},$$
 (4)

$$X_{i,j} \le y_i, \forall j \in \{1, ..., N\}, \forall i \in \{1, ..., M\},$$
(5)

$$\sum_{i=0}^{M} X_{i,j} = 1, \forall j \in \{1, ..., N\},$$
(6)

$$X_{i,j} \ge 0, \forall i \in \{0, ..., M\}, \forall j \in \{1, ..., N\}, \text{ and}$$
 (7)

$$K = \sum_{i=1}^{M} \sum_{j=1}^{N} n_j X_{i,j} .$$
(8)

 $y_i$  are binary variables equal to 1 if extended gate *i* is open and 0 otherwise. Constraint (3) ensures that, at most, one extended gate can be opened. Constraints (5) imply that flow can be routed only through an open extended gate. Constraints (6) ensure that the total amount of flow is shipped from origin to destinations. Constraints (7) ensure that the proportions of flow routed are non-negative. Finally, constraint (8) is used to account for the number of containers routed by intermodal transportation.

Objective function (2) with constraints (3) - (8) is the classical hub location problem with flowdependent economies of scale. This corresponds to the single actor view of the problem, i.e., when all the destinations align their decisions in order to minimize the total transportation and transshipment cost (SO allocation) and when the terminal operator also decides to locate the extended gate in order to minimize the total cost. This problem serves as a benchmark for assessing the impact of having multiple actors involved.

We consider several variations of this problem. First, we consider that the destinations act independently and seek to minimize their own transportation and transshipment cost. Second, we consider another objective for locating the extended gate by considering that the terminal operator seeks to maximize the terminal utilization. This second objective function may be expressed as follows:

### MAX K.

(9)

The different variations are all solved by backward induction. First, we focus on the allocation decisions made by the destinations (i.e., the followers of the game) and we assume that the extended gate  $i \in \{1,...,M\}$  is open. Second, the optimal extended gate location for the two objective functions considered is obtained by enumeration. Section 4 focuses on the first step of the backward induction procedure, i.e., on the allocation problem. After stating some preliminary results in Subsection 4.1, Subsection 4.2 focuses on identifying all the existing user equilibria for the shippers' allocation game. The SO allocation is then identified in Subsection 4.3. The full procedure is illustrated via an example in Section 5.

### 4. The shippers' allocation problem

In this section, we aim at solving the shippers' allocation problem (i.e., the problem of deciding how to allocate the demand flow between direct shipment and intermodal transportation for each destinations). We assume that an extended gate  $i \in \{1, ..., M\}$  is open and we focus on the routing

decisions made by the destinations. The destinations indeed need to determine which share of their demand flow is routed via the extended gate *i* (i.e., determine  $X_{i,j}$ ), knowing that the remaining share of their demand flow will be routed by direct shipment from the port ( $X_{0,j} = 1 - X_{i,j}$ ). After stating a necessary condition for obtaining a user equilibrium and a system optimal allocation that enables simplifying the analysis, we identify the user equilibria first and then the system optimal allocation.

### 4.1 Preliminary result

We start by formally defining a user equilibrium (UE). As Fisk (1984) points out, Wardrop's first principle of user equilibrium is related to the Nash equilibrium principle in non-cooperative game theory. Indeed, Haurie and Marcotte (1985) show that a Nash equilibrium in an atomic congestion game converges to a Wardrop equilibrium when the number of players increases. In the setting we consider, each destination (i.e., each shipper) is considered a player of the game. Each destination  $j \in \{1, ..., N\}$  aims at minimizing its total cost function expressed as follows:

$$Z_{i,j}(X_{i,j},K) = n_j \Big( (1 - X_{i,j}) Z_{0,j}^1 + X_{i,j} (Z_{0,i}^3(K) + Z_{i,j}^1) \Big).$$
(10)

A Nash equilibrium is obtained when neither player can unilaterally reduce its transportation cost by changing its decision. Nash (1950) proves that there is at least one Nash equilibrium under general assumptions by applying a fixed point theorem. However, this equilibrium may be obtained by requiring at least one player to choose a probability distribution over the set of potential actions to protect against other players' reactions. This type of strategy is a mixed strategy, as opposed to a pure strategy. A pure strategies Nash equilibrium is such that each player chooses an action for sure. This type of equilibrium may not always exist (see e.g., rock-paper-scissors or matching pennies games).

We can notice that even though the cost function (10) does not depend on the other players' decisions directly, it does so via the number of containers that will be shipped through the extended gate. This feature is consistent with atomic congestion games as defined by Rosenthal (1973), except that in our setting, the more intermodal transportation is used, the better. Rosenthal (1973) proves that there exists at least a pure strategies Nash equilibrium for atomic congestion games. We may expect that a pure strategies Nash equilibrium exists as well in our setting and we focus only on pure strategies Nash equilibria in what follows. In order to fit to shippers' practices, we additionally assume that the extended gate is used by a destination only if this leads to a strict decrease in cost compared to direct shipment. We consequently define a user equilibrium (UE) as a pure strategy Nash equilibrium such that the players choosing to use the extended gate are strictly better off in doing so, compared to using direct shipment only. We show in the next section that a UE always exists for the shippers' allocation game, and that several UE may exist in some situations.

We define a system optimum (SO) allocation as an allocation minimizing the total cost with the minimum volume shipped via the extended gate. We show in subsection 4.3 that this definition ensures the uniqueness of the SO allocation, due to the structural properties of the problem.

 $X_{i,j}$  satisfies the single routing condition if and only if  $X_{i,j} \in \{0,1\}$ , i.e., if the entire flow of destination *j* is routed either by direct shipment or by intermodal transportation via extended gate *i*. Theorem 1 states that the single routing condition is satisfied for each destination under any UE (if any) and under any SO allocation (if any).

**Theorem 1:** Any UE and any SO allocation satisfies the single routing condition for all  $j \in \{1, ..., N\}$ .

All proofs appear in Appendix B (available online as supplementary material). We recall here that we exclude urgent deliveries from the model as the unplanned share of the demand will necessarily be shipped directly from the port and consequently this does not influence the decisions made and the results obtained. Taking urgent deliveries into account, a destination may be delivered both by direct shipment (for urgent deliveries) and by intermodal transportation (for non-urgent deliveries) in practice. Theorem 1 states that the single routing condition holds for the non-urgent share of the demand at destination, i.e., the quantity considered in the model. Theorem 1 also implies that a SO allocation exists for all  $i \in \{1,...,M\}$ , as  $\{0;1\}^N$  is a finite set.

#### 4.2 Identifying the user equilibria of the game

In this subsection, we prove that a UE always exists and that several user equilibria may exist for some instances of the shippers' allocation game. Then, we identify key structural properties that enable identifying the existing equilibria. Finally, we define the effort that the terminal operator needs to exert to obtain a given UE by characterizing the minimum number of destinations that need to be convinced to get the base volume necessary for becoming competitive and we show how to compute this effort.

We start by turning the cost function for each destination as expressed in Formula (10) into a profit function by computing the cost savings of a strategy compared to using only direct shipment:

$$P_{i,j}(X_{i,j},K) = X_{i,j}n_j \Big( Z_{0,j}^1 - Z_{i,j}^1 - Z_{0,i}^3(K) \Big).$$
<sup>(11)</sup>

By applying Theorem 1,  $X_{i,j} \in \{0,1\}$  under any UE. Consequently, under any UE, destination *j* will use the extended gate if and only if  $P_{i,j}(1,K) > 0$  (by definition of a UE, we set  $X_{i,j} = 0$  if  $P_{i,j}(1,K) = 0$ ). As,  $P_{i,j}(1,K)$  depends on *K* (the total number of container shipped by train), we define the minimum volume required to make intermodal transportation via extended gate *i* economically viable for destination *j* as  $K^{i,j}$  (we recall that we consider *i* as fixed in this section).  $K^{i,j}$  is the smallest integer value of *K* such that  $P_{i,j}(1,K) > 0$ . If  $K^{i,j}$  exists, then it is unique. If  $K^{i,j}$  does not exist, then  $X_{i,j} = 0$  independently of the decisions of the other destinations and we set  $K^{i,j} = \infty$ . Theorem 2 proves that at least one UE exists for the shippers' allocation game and Theorem 3 states that the UE may not always be unique.

**Theorem 2:** The shippers' allocation game possesses at least one user equilibrium.

#### **Theorem 3:** The shippers' allocation game may possess several user equilibria.

As we show in the proof (see Appendix B in the supplementary material), Theorem 3 holds even when considering only two destinations. This occurs when neither of the destinations has enough volume to individually make use of the extended gate and when the combination of both destinations' volumes makes the use of the extended gate profitable for each. This example can be viewed as a stag hunt game (Skyrms, 2001) for which two pure strategy Nash equilibria exist. In what follows, we develop structural properties of the shippers' allocation game. These properties enable us to identify all the existing user equilibria for the game.

 $K^{i,j}$  also enables us to define the conditions to reach a UE. Let  $L_i$  be the number of user equilibria of the shippers' allocation game associated to extended gate i ( $L_i \ge 1$  by applying Theorem 2). For all  $l \in \{1,...,L_i\}$ , let  $U_{i,l}$  be the set of destinations using the extended gate (such that  $X_{i,j} = 1$ ) under equilibrium l ( $U_{i,l}$  may be empty). Then,  $U_{i,l}$  satisfies the following conditions:

$$\max_{j \in U_{i,l}} (K^{i,j}) \le \sum_{j \in U_{i,l}} n_j, \qquad (12)$$

$$K^{i,t} > \sum_{j \in U_{i,l}} n_j + n_i, \ \forall t \in \{1, ..., N\} \setminus U_{i,l}.$$
(13)

Condition (12) implies that the volume generated by the destinations included into  $U_{i,l}$  is large enough to have all these destinations using the extended gate. Conditions (13) imply that the remaining destinations are not able to individually join the set of destinations using the extended gate. The shippers' allocation game highlights nice structural properties, stated in Theorem 4 and Corollaries 1, 2 and 3.

**Theorem 4:** Let  $a, b \in \{1, ..., N\}$  be two distinct destinations such that  $K^{i,b} \ge K^{i,a}$ , then:  $b \in U_{i,l} \Rightarrow a \in U_{i,l}$ , for all  $l \in \{1, ..., L_i\}$ .

**Corollary 1:** Consider an instance of the shippers' allocation game with  $L_i \ge 2$  user equilibria. Then, it is possible to arrange the  $L_i \ge 2$  user equilibria such that:

$$U_{i,l} \subset U_{i,l+1}, \forall l \in \{1, ..., L_i - 1\}.$$

**Corollary 2:** Consider an instance of the N shippers' allocation game with  $L_i$  user equilibria. Then:

$$L_i \leq \left\lfloor \frac{N}{2} \right\rfloor + 1$$

**Corollary 3:** Consider an instance of the shippers' allocation game with  $L_i \ge 2$  user equilibria. Then, for all  $l \in \{1, ..., L_i - 1\}$ ,  $U_{i,l+1}$  performs strictly better than  $U_{i,l}$  in terms of extended gate utilization and in terms of total cost.

Theorem 4 enables simplifying the search for the user equilibria by arranging the destinations by increasing order of  $K^{i,j}$ . We assume w.l.o.g. for the remaining of this subsection that the destinations are arranged such that  $K^{i,1} \leq ... \leq K^{i,M}$ . Conditions to reach a UE with the first k  $(1 \leq k < N)$  destinations are expressed as follows:

$$K^{i,k} \le \sum_{j \in \{1,\dots,k\}} n_j , \qquad (14)$$

$$K^{i,t} > \sum_{j \in \{1,...,k\}} n_j + n_t, \ \forall t \in \{k+1,...,N\}.$$
(15)

The conditions to reach a UE with no destination using the extended gate are:

$$K^{i,t} > n_i, \ \forall t \in \{1, \dots, N\},\tag{16}$$

and the condition to reach a UE with all the destination using the extended gate is:

$$K^{i,N} \le \sum_{j \in \{1,\dots,N\}} n_j . \tag{17}$$

We apply these conditions in the example presented in Table 1, with 10 destinations. The UE identified are  $U_{i,2} = \{1;2;3;4;5\}$  by applying Conditions (14) and (15), as well as  $U_{i,1} = \emptyset$  by applying Condition (16). Note that  $\{1;2\}$  satisfies Condition (14). However,  $K^5 < \sum_{j \in \{1,2\}} n_j + n_5$ 

(i.e.,21<16+6), which contradicts Condition (15).

destination j	n <sub>j</sub>	K <sup>i,j</sup>	∑n <sub>j</sub>
1	7	9	7
2	9	14	16
3	2	19	18
4	2	20	20
5	6	21	26
6	4	31	30
7	5	50	35
8	15	62	50
9	12	8	62
10	6	8	68

Table 1: An example of the 10 shippers' allocation game

Corollary 2 states that the maximum number of UE for a given instance of the *N* shippers' allocation game grows linearly in *N*. As the conditions (14) - (17) can be checked in polynomial time in *N*, we can identify all the UE for the *N* shippers' allocation game in polynomial time of *N*. This ensures practical application.

Our results show that several user equilibria may exist for the shippers' allocation game. We also show how to identify the existing set of UE. To our knowledge, the only models considering UE in an intermodal hub location problem are proposed by Yamada et al. (2009) and by Meng and Wang (2011). In their analysis, the total transportation cost is minimized under constraints stating that a destination may not be routed via a path that increases its individual transportation cost compared to direct shipment. Thus, these two articles focus on identifying the UE that minimizes the total cost (i.e.,  $U_{i,L_i}$  by applying Corollary 3). Our results question this choice of focusing solely on the UE minimizing the total cost as we prove that several equilibria may exist. Corollary 3 shows that the terminal operator would favor  $U_{i,L_i}$  as this equilibrium performs better for the two objectives of interest for the terminal operator. However, we cannot ensure that  $U_{i,L_i}$  is the equilibrium that would be found in practice as soon as  $L_i > 1$ .

In what follows, we consequently propose to identify the minimal number of destinations who have to use the extended gate to ensure that the total cost given by objective function (2) and the extended gate utilization given by objective function (9) are at least as good as the ones obtained under  $U_{i,l}$ ,  $l \in \{1,...,L_i\}$ . Indeed, the terminal operator often needs to convince some shippers to use the extended gate in order to obtain the base volume necessary for becoming competitive. Let Card(U)be the cardinality of a set U. We propose and use the following definition of a set compatible with  $U_{i,l}$  to identify the set of shippers who are able to generate this base volume.

**Definition 1:** Let  $U_{i,l}$ ,  $l \in \{1,...,L_i\}$  be a user equilibrium. A set U is called compatible with  $U_{i,l}$  if this one satisfies the following conditions:

- $U \subseteq U_{i,l}$ ,
- $\quad \quad \forall k \in U\,, \ K^{i,k} \leq \sum_{j \in U} n_j\,,$
- If  $U_{i,l} \setminus U \neq \emptyset$ , there exists a permutation of the elements of  $U_{i,l} \setminus U$  denoted as  $(1,...,Card(U_{i,l}) Card(U))$  such that for all  $k \in (1,...,Card(U_{i,l}) Card(U))$ ,  $K^{i,k} \leq \sum_{j \in U} n_j + \sum_{j=1}^k n_k$ .

We can first notice that if all the destinations of a set U compatible with  $U_{i,l}$  use the extended gate, the volume generated ensures that each of these destinations are better off as compared to using direct shipment. In addition, the destinations of  $U_{i,l} \setminus U$  (if any) can iteratively join the extended gate, by following the sequence of the permutation  $(1,...,Card(U_{i,l}) - Card(U))$ . Consequently, if the terminal operator manages to convince all the destinations included in a set U compatible with  $U_{i,l}$  to use the extended gate, the total cost given by objective function (2) and the extended gate utilization given by objective function (9) will be at least as good as the ones obtained under  $U_{i,l}$ . Identifying a set compatible with  $U_{i,l}$  in practice is difficult as this implies identifying the sequence that enables destinations of  $U_{i,l} \setminus U$  to iteratively join the extended gate. Theorem 5 enables simplifying the search for the sets compatible with  $U_{i,l}$ .

**Definition 2:** Let  $U_{i,l}$ ,  $l \in \{1, ..., L_i\}$  be a user equilibrium. Let  $U_{i,l}^{\min}$  be the set of all sets U satisfying the following conditions:

- $U \subseteq U_{i,l}$ ,
- $\quad \quad \forall k \in U\,, \ K^{i,k} \leq \sum_{j \in U} n_j\,,$

- *if* 
$$l > 1$$
,  $U \not\subset U_{i,l-1}$ .

**Theorem 5:** Let  $U_{i,l}$ ,  $l \in \{1, ..., L_i\}$  be a user equilibrium. Let  $U_{i,l}^{comp}$  be the set of all sets compatible with  $U_{i,l}$  and let  $U_{i,l}^{\min}$  be the set of sets defined in Definition 2:

Then, 
$$U_{i,l}^{\min} = U_{i,l}^{comp}$$

Theorem 5 enables easily verifying if a given set is compatible with  $U_{i,l}$ . We further define the minimal number of destinations who need to be convinced to obtain performances as least as good as the ones obtained under  $U_{i,l}$ . First, we can notice that  $U_{i,l}^{comp}$  is non-empty as  $U_{i,l}$  is compatible with  $U_{i,l}$ . In addition,  $U_{i,l}^{comp}$  is a finite set as there is a finite number of subsets of a finite set  $(U_{i,l})$  is a finite set). We can consequently define  $Eff(U_{i,l}) = \min_{U \in U_{i,l}^{comp}} (Card(U))$ .  $Eff(U_{i,l})$  corresponds to the effort that the terminal operator needs to exert to ensure obtaining the performances of  $U_{i,l}$ . In practice, we do not necessarily need to identify all the sets compatible with  $U_{i,l}$  to compute

 $Eff(U_{i,l})$ . Indeed, if we identify a set U compatible with  $U_{i,l}$  such that there is no set compatible with  $U_{i,l}$  with cardinality less or equal to Card(U)-1, then  $Eff(U_{i,l}) = Card(U)$ .

In the example proposed above with 10 destinations, we obtain that  $Eff(U_{i,1}) = 0$  (as  $U_{i,1} = \emptyset$ ) and  $Eff(U_{i,2}) = 2$  (as  $\{1,2\}$  is compatible with  $U_{i,2}$ , and there is no singleton satisfying  $K^{i,j} \le n_j$  as  $U_{i,1} = \emptyset$ ). We compute  $Eff(U_{i,l})$  is Section 5 and we further analyze the impacts of having more than one user equilibrium and derive several insights.

### 4.3 System optimum allocation and price of stability

The price of stability associated to the extended gate *i* for the *N* shippers' allocation game with  $L_i \ge 1$  user equilibria is defined as the ratio between the total cost obtained under  $U_{i,L_i}$  and the minimum cost that could be obtained. This concept has firstly been proposed by Anshelevich et al. (2008). This is an optimistic evaluation of the consequences of having several actors interacting in the *N* shippers' allocation game. Indeed,  $U_{i,L_i}$  is the user equilibrium that performs the best in terms of total cost according to Corollary 3. The price of stability associated to the extended gate *i* is referred to as  $PoS_i$  (note that  $PoS_i \ge 1$  for all  $i \in \{1, ..., M\}$ ). The price of stability is the optimistic version of the price of stability to measure the value of collaboration in what follows.

The computation of the price of stability involves identifying a SO allocation, as defined in subsection 4.1. Under a SO allocation, some destinations may decide to use the extended gate even if this choice increases their individual cost, as soon as the volume added for train transportation and transshipment operations provides a stronger cost reduction for the other players already using the extended gate. Let  $S_i$  be the set of all players using the extended gate under a SO allocation ( $S_i$  may be empty). Theorem 6 states the key structural property of  $S_i$ .

**Theorem 6:** Assume that there exist two distinct destinations  $a, b \in \{1, ..., N\}$  such that  $Z_{0,a}^1 - Z_{i,a}^1 \ge Z_{0,b}^1 - Z_{i,b}^1$ , then:

$$b \in S_i \Longrightarrow a \in S_i$$
.

Theorem 6 implies that ranking and relabeling the destinations  $j \in \{1,...,N\}$  by decreasing order of  $Z_{0,j}^1 - Z_{i,j}^1$  helps in finding the SO allocation. We assume w.l.o.g. for the remaining of this subsection that the destinations are arranged such that  $Z_{0,1}^1 - Z_{i,1}^1 \ge ... \ge Z_{0,N}^1 - Z_{i,N}^1$ . The function obtained by

comparing the total cost incurred when the first k destinations use the extended gate and the total cost incurred when all the destinations use direct truck transportation can be expressed as follows:

$$\sum_{j=1}^{k} n_j (Z_{0,j}^1 - Z_{i,j}^1) - \sum_{j=1}^{k} n_j Z_{0,i}^3 \left( \sum_{j=1}^{k} n_j \right).$$
(18)

This profit function can be evaluated for all  $0 \le k \le N$  and  $S_i$  is obtained by selecting the highest value obtained with the minimum number of destinations using intermodal transportation. Accordingly, Theorem 6 implies the uniqueness of the SO allocation as defined in Subsection 4.1. Note that the profit function as evaluated by Formula (18) may not always be a monotonic function of k. The following Lemma shows that ordering the destinations by decreasing order of  $Z_{0,j}^1 - Z_{i,j}^1$ implies ordering the destinations by increasing order of  $K^{i,j}$ .

**Lemma 1:** Assume that the destinations are ordered such that  $Z_{0,j}^1 - Z_{i,j}^1 \ge Z_{0,j+1}^1 - Z_{i,j+1}^1$  for all  $j \in \{1, ..., N-1\}$ , then:

$$K^{i,j} \leq K^{i,j+1}$$
, for all  $j \in \{1,..., N-1\}$ .

Finally, Lemma 2 enables comparing the set of destinations using the extended gate under the SO allocation and the set of destinations using the extended gate under  $U_{iL}$ .

**Lemma 2:** Let  $S_i$  be the set of destinations using extended gate *i* under the system optimum allocation and let  $U_{i,L_i}$  be the set of destinations using the extended gate under the user equilibrium performing the best in terms of total cost, then:

$$U_{i,L_i} \subseteq S_i$$
.

Lemma 2 implies that  $PoS_i > 1$  if and only if  $U_{i,L_i} \neq S_i$ . Lemma 2 also implies that the SO allocation performs at least as good at  $U_{i,L_i}$  in terms of extended gate utilization.

### 5. Example and insights

This section is based on an example representing features of the hinterland network in the Netherlands. The main objective here is to explore the implications of having multiple actors involved in such a supply chain. We apply the modeling developed in Section 3, and the theoretical results of Section 4 enable us to quickly solve the problem. In this example, N = 25 and M = 10. More information about the data used for this example may be found in Appendix C (available online as supplementary material).

We determine  $U_{i,l}$  and  $Eff(U_{i,l})$ , for all  $i \in \{1,...,M\}$  and for all  $l \in \{1,...,L_i\}$ . This allows us to compute the total cost for all  $U_{i,l}$  as well as the extended gate utilization by dividing the actual number of containers shipped to the extended gate by the total amount of containers (i.e.,  $K / \sum_{j=1}^{N} n_j$ ). The results appear in Table 2. Note that in case the extended gate is not used, the total transportation cost is  $\notin 1274$ .

	Total Cost (€)	Extended Gate Utilization (%)	$Eff(U_{i,l})$
U1,1	11 274	0%	0
U1,2	10 410	30%	5
U1,3	9 565	43%	10
U2,1	11 274	0%	0
U2,2	9 802	48%	12
U3,1	11 274	0%	0
U4,1	11 274	0%	0
U4,2	11 070	22%	6
U5,1	11 274	0%	0
U5,2	10 653	24%	6
U6,1	11 274	0%	0
U6,2	10 676	35%	10
U7,1	11 274	0%	0
U7,2	11 197	28%	6
U7,3	10 908	57%	9
U8,1	11 274	0%	0
U8,2	9 084	52%	6
U9,1	11 274	0%	0
U10,1	11 274	0%	0

Table 2: The user equilibria for an example with 25 shippers and 10 extended gate locations

We can notice that several equilibria exist for most of the potential extended locations considered. Out of 10 potential locations, 7 possess several UE. In addition, a user equilibrium involves having no destination using the extended gate for all the potential locations considered.

**Insight 1:** Several user equilibria often exist in practice. In most of the cases, one of these equilibria involves not using the extended gate.

Insight 1 is consistent with the practical findings of the maritime economics and transport geography literature. Indeed, Van Der Horst and De Langen (2008) mention that introducing a new hinterland service requires a base volume for becoming competitive. This base volume may often not be met by a single shipper. In case the base volume is not met, the user equilibrium will consist of having no destination using the extended gate. Accordingly, we evaluate the minimal number of destinations that the terminal operator needs to convince to ensure that at least the level of performance provided

under  $U_{i,l}$  is obtained by computing  $Eff(U_{i,l})$ . None of the user equilibria with at least one destination using the extended gate may be obtained without convincing at least 5 destinations for the presented example, as shown in Table 2.

The different user equilibria may be evaluated by the terminal operator based on the total cost (to be minimized), the extended gate utilization (to be maximized) and the convincing effort measured by  $Eff(U_{i,l})$  (to be minimized). We apply multiobjective optimization to identify the set of efficient user equilibria. A user equilibrium  $U_{i,l}$  is said to be efficient if none of the other equilibria perform at least as good at  $U_{i,l}$  for two of the objectives considered and strictly better than  $U_{i,l}$  for the remaining objective. Note that user equilibria obtained for different locations of the extended gate can be compared by the terminal operator as this one decides on where to locate the extended gate. The literature on combinatorial multiobjective optimization is well developed and enables implementing reliable and fast methods to identify the efficient solutions (see e.g., Ehrgott, 2005; Phelps and Köksalan, 2003). For the example of interest here, the efficient user equilibria may be obtained by direct comparison as the number of user equilibria is limited. The results appear in Table 3. We additionally define  $\delta_{0,i}$  as the distance from the deep-sea terminal to the extended gate (expressed in km).

i	Total Cost (€)	Extended Gate Utilization (%)	$Eff(U_{i,l})$	$\delta_{\scriptscriptstyle 0,i}$ (km)
N/A	11 274	0%	0	N/A
1	10 410	30%	5	197
8	9 084	52%	6	182
7	10 908	57%	9	31

Table 3: The efficient user equilibria

Based on the information provided in Table 3, the terminal operator can decide on a targeted total cost and extended gate utilization, select the best extended gate location accordingly and convince the destinations in accordance with  $Eff(U_{i,l})$  (note that the destinations who need to be convinced appear

in  $\underset{U \in U_{i,i}^{comp}}{\operatorname{arg\,min}}(Card(U))$ ). Table 3 highlights that the convincing effort that the terminal operator is

willing to make influences the optimal extended gate location.

**Insight 2:** The optimal extended gate location depends on the effort that the terminal operator is willing to make.

Insight 2 implies that the terminal operator needs to assess the convincing effort and the outcome of this effort (if the destinations will or will not be convinced) in order to choose the optimal extended gate location. The question of how to convince destinations to use the extended gate in practice is of great interest. For example, the destinations may agree on guaranteed minimum volumes shipped via the extended gate before implementing this one. Such practice is currently used in the Netherlands, where 11 shippers in the region of Westland have signed an agreement to transport 10 000–15 000 containers per year by barge from the port of Rotterdam to the container terminal of Hoek of Holland (project Fresh Corridor 7). In this example, the project is not led by a terminal operator but we may expect such experience to be relevant for extended gate projects as well. We refer to the maritime economics literature focusing on hinterland supply chains for related discussions (see, e.g., Van Der Horst and De Langen, 2008).

Table 3 also shows that the total cost and extended gate utilization may be conflicting objectives. Indeed, minimizing the total cost and maximizing the extended gate utilization do not lead to the same location decisions for the example studied here.

# **Insight 3:** The classical objective of minimizing the total cost cannot be used as a proxy for maximizing the extended gate utilization as these two objectives may be conflicting.

Insight 3 is somehow counterintuitive as we might expect that lowering the total cost would encourage more destinations to use the extended gate. However, the distance from the port to terminals helps understanding the dynamic. This information is presented in Table 3. Note that the distances are quite low for the example considered, as the example represents features of the hinterland in the Netherlands. Higher distances are expected if the results are applied to another area. We can notice that the extended gate utilization is decreasing in the distance from the port. Indeed, the more far away the extended gate is from the port, the more destinations are located closer to the port than to the extended gate. If destination j is located closer to the port than to extended gate i, we may expect that  $Z_{0,i}^1 < Z_{i,i}^1$ , leading to  $K^{i,j} = \infty$ . Thus this destination would not use the extended gate (independently of the volume shipped to the extended gate). In addition, we can expect that the train transport cost is increasing in  $\delta_{0,i}$ , thus the minimal volume required to have the destinations using the extended gate tends to be increasing in the extended gate distance from the port. As a result, the extended gate utilization is expected to be higher when this one is located closer to the port. On the other hand, midrange locations perform better in terms of total cost as extended gate utilization is still reasonable and the distance from the port to the extended gate enables efficient use of intermodal transportation.

**Insight 4:** *Extended gates close to the port perform better in terms of utilization and extended gate at midrange locations perform better in terms of total cost.* 

As Roso et al. (2009) point out, inland terminals may be located at different distances from the port when fulfilling different functions. Our results suggest to assess the comparative advantages and drawbacks of inland terminal locations in terms of total cost and utilization. We can also expect extended gates to be located closer to the port compared to other types of inland terminals as we highlight in Section 3 that utilization is of primary importance for terminal operators.

In order to fully explore the implications of having multiple actors involved in intermodal hinterland supply chains, we determine  $S_i$  and  $PoS_i$  for all  $i \in \{1,...,M\}$ . We recall here that the price of stability  $PoS_i$  is the ratio between the cost obtain under  $U_{i,L_i}$  and the cost obtain under  $S_i$ . This one enables us to evaluate the value of collaboration. The results appear in Table 4.

	Total Cost (€)	Extended Gate Utilization (%)	$PoS_i$
<b>S1</b>	9 272	54%	1,03
S2	9 579	61%	1,02
<b>S</b> 3	11 186	41%	1,01
<b>S4</b>	10 580	74%	1,05
S5	9 897	52%	1,08
<b>S6</b>	9 936	57%	1,07
S7	10 846	70%	1,01
<b>S8</b>	8 946	63%	1,02
S9	10 112	67%	1,11
S10	10 309	57%	1,09

Table 4: The System Optimum allocations

The results of Table 4 enable us to draw additional insights. First, we can notice from Table 4 that the price of stability is relatively low for all the locations of the example. On the other hand, collaboration is compulsory for getting the extended gate used for locations 3, 9 and 10. This means that collaboration is critical for implementing the extended gate in some cases. The results obtained for location 3 are particularly surprising. For this location, there is a unique UE consisting in not using the extended gate (see Table 2). On the other hand the SO allocation lead to an extended gate utilization of 41%. We might expect the value of collaboration to be quite high (compared to other locations) in such situation. However, the value of collaboration is very low for location 3.

**Insight 5:** Collaboration may be critical for successfully implementing an extended gate even if the value of collaboration is generally quite low.

Finally, our results can help explain why some intermodal transportation projects are predicted to be effective in theory while being very difficult to turn into profitable projects in practice. This statement

is in accordance with Rodrigue et al. (2010), who report that "both public and private actors have a tendency to overestimate the benefits and traffic potential and underestimate the costs and externalities of inland port projects" (p. 528). Most of the models used for estimating the benefits and traffic potential of inland terminal projects neglect coordination issues. The setting corresponds to the SO allocation of the proposed model. Lemma 2 implies that the SO allocation performs better than any UE in terms of total cost and extended gate utilization in most of practical situations. This enables understanding why public and private actors tends to overestimate the benefits and traffic potential in practice. The impacts of this overestimation can be substantial. For locations 3, 9 and 10, the SO allocation lead to a solution that may seem appealing, while the only existing UE has no destinations using the extended gate.

**Insight 6:** *The overestimation in extended gate utilization resulting from ignoring that multiple actors interact can lead to misleadingly consider an extended gate project as profitable.* 

Our results lead us to conclude that taking the multiple actors feature of intermodal hinterland networks into account is of primary importance, especially because no single actor usually fulfills the role of supply chain leader in hinterland supply chains (Bontekoning et al., 2004).

### 6. Conclusions

In this article, we analyze some implications of having multiple actors involved in intermodal hinterland supply chains by focusing on a new problem referred to as the extended gate location problem. In this problem, the terminal operator first decides on where to locate the extended gate. Then the shippers decide if they want to take their containers at the deep-sea terminal or at the extended gate. We formulate the shippers' allocation problem as a non-cooperative game and we derive key structural properties of the game. This allows us to identify all the existing equilibria for the game. In addition, we show how to compute the minimum number of shippers who need to be convinced to reach a given equilibrium and we measure the value of collaboration by comparing our results to the ones obtained for the shippers' allocation leading to the optimal cost. We apply the results to an example based on the features of the hinterland network in the Netherlands and provide related insights.

Our main findings are as follows. First, we show that several equilibria often exist in practice. In most of the cases, the terminal operator needs to convince some shippers to use the extended gate in order to obtain the base volume necessary for becoming competitive. Second, we show that the optimal location depends on the convincing effort the terminal operator is willing to make. Third, we highlight that the minimization of the total cost and the maximization of the extended gate utilization may be conflicting objectives and we propose to apply multiobjective optimization to identify the set

of efficient solutions. Finally, we highlight that collaboration may be critical for successfully implementing an extended gate even if the value of collaboration may also be quite low. Overall, we prove that the multiple actors feature of intermodal hinterland networks is critical and needs to be accounted for. We focus on the extended gate location problem in this article but our results may serve as a basis for appropriately taking multiple actors into account in other hinterland hub location problems. The results may also generalize to the entire container supply chain, but further research is necessary for this end.

A natural extension of the article would consist in studying the problem of opening multiple extended gates. This problem is an adaptation of the p-hub median problem known to be NP hard. We expect some of the structural properties demonstrated for the single extended gate location problem to hold when considering multiple extended gates. If our expectations materialize, the methods developed for solving the single actor p-hub median problem could then be adapted. A second way of extending this article would consist in studying how to provide incentives to the shippers, such that they act in accordance with the system optimum solution. This may be done by proposing rebates to some shippers in case they use the extended gate. We refer to Maillé and Stier-Moses (2009) for an illustration on how rebates may be used in the traffic assignment literature. The shippers may also seek for cooperation. In this case, the shippers' allocation problem may be viewed as a cooperative game and the key question consists in identifying stable cost allocations (see e.g., Özener and Ergun, 2008). We hope that our results help pave the way for further research from the operations management and transportation science community on container transportation systems.

### Acknowledgments

The research was supported by Dinalog, the Dutch Institute for Advanced Logistics.

### References

- Agarwal, R., Ergun, Ö., 2008. Ship Scheduling and Network Design for Cargo Routing in Liner Shipping. Transportation Science 42, 175–196.
- Alumur, S.A., Kara, B.Y., Karasan, O.E., 2012a. Multimodal hub location and hub network design. Omega 40, 927–939.
- Alumur, S.A., Yaman, H., Kara, B.Y., 2012b. Hierarchical multimodal hub location problem with time-definite deliveries. Transportation Research Part E: Logistics and Transportation Review 48, 1107–1120.
- Alumur, S., Kara, B.Y., 2008. Network hub location problems: The state of the art. European Journal of Operational Research 190, 1–21.
- Anshelevich, E., Dasgupta, A., Kleinberg, J., Tardos, É., Wexler, T., Roughgarden, T., 2008. The Price of Stability for Network Design with Fair Cost Allocation. SIAM J. Comput. 38, 1602– 1623.
- Arnold, P., Peeters, D., Thomas, I., 2004. Modelling a rail/road intermodal transportation system. Transportation Research Part E: Logistics and Transportation Review 40, 255–270.
- Bontekoning, Y., Macharis, C., Trip, J., 2004. Is a new applied transportation research field emerging?—A review of intermodal rail-truck freight transport literature. Transportation Research Part A: Policy and Practice 38, 1–34.
- Campbell, J.F., O'Kelly, M.E., 2012. Twenty-Five Years of Hub Location Research. Transportation Science 46, 153–169.
- Crainic, T.G., Kim, K.H., 2007. Intermodal Transportation, in: Transportation. Volume 14 of Handbooks in Operations Research and Management Science, pp. 467–537.
- De Langen, P.W., Chouly, A., 2004. Hinterland Access Regimes in Seaports. European Journal of Transport and Infrastructure Research.
- Ehrgott, M., 2005. Multicriteria Optimization. Birkhäuser.
- Farahani, R.Z., Hekmatfar, M., Arabani, A.B., Nikbakhsh, E., 2013. Hub location problems: A review of models, classification, solution techniques, and applications. Computers & Industrial Engineering 64, 1096–1109.
- Fisk, C.S., 1984. Game theory and transportation systems modelling. Transportation Research Part B: Methodological 18, 301–313.
- Fransoo, J.C., Lee, C.-Y., 2013. The Critical Role of Ocean Container Transport in Global Supply Chain Performance. Production and Operations Management 22, 253–268.
- Groothedde, B., Ruijgrok, C., Tavasszy, L., 2005. Towards collaborative, intermodal hub networks: A case study in the fast moving consumer goods market. Transportation Research Part E: Logistics and Transportation Review 41, 567–583.
- Haurie, A., Marcotte, P., 1985. On the relationship between Nash—Cournot and Wardrop equilibria. Networks 15, 295–308.
- Ishfaq, R., Sox, C.R., 2012. Design of intermodal logistics networks with hub delays. European Journal of Operational Research 220, 629–641.
- Ishfaq, R., Sox, C.R., 2011. Hub location–allocation in intermodal logistic networks. European Journal of Operational Research 210, 213–230.
- Ishfaq, R., Sox, C.R., 2010. Intermodal logistics: The interplay of financial, operational and service issues. Transportation Research Part E: Logistics and Transportation Review 46, 926–949.
- Jeong, S.-J., Lee, C.-G., Bookbinder, J.H., 2007. The European freight railway system as a hub-andspoke network. Transportation Research Part A: Policy and Practice 41, 523–536.

- Koutsoupias, E., Papadimitriou, C., 1999. Worst-Case Equilibria, in: Meinel, C., Tison, S. (Eds.), STACS 99, Lecture Notes in Computer Science. Springer Berlin Heidelberg, pp. 404–413.
- Levinson, M., 2010. The Box: How the Shipping Container Made the World Smaller and the World Economy Bigger. Princeton University Press.
- Limbourg, S., Jourquin, B., 2009. Optimal rail-road container terminal locations on the European network. Transportation Research Part E: Logistics and Transportation Review 45, 551–563.
- Maillé, P., Stier-Moses, N.E., 2009. Eliciting Coordination with Rebates. Transportation Science 43, 473–492.
- Meng, Q., Wang, X., 2011. Intermodal hub-and-spoke network design: Incorporating multiple stakeholders and multi-type containers. Transportation Research Part B: Methodological 45, 724–742.
- Nash, J.F., 1950. Equilibrium points in n-person games. PNAS 36, 48-49. doi:10.1073/pnas.36.1.48
- Notteboom, T.E., 2008a. Bundling of freight flows and hinterland network developments, in: The Future of Intermodal Freight Transport. Edward Elgar Publishing.
- Notteboom, T.E., 2008b. The relationship between seaports and the intermodal hinterland in light of global supply chains: European challenges (No. disucssion paper No. 2008-10). OECD International Transport Forum.
- Notteboom, T.E., Rodrigue, J.-P., 2005. Port regionalization: towards a new phase in port development. Maritime Policy & Management 32, 297–313.
- OECD/ITF, 2009. Port competition and hinterland connections. Round Table 143, OECD Publishing.
- O'Kelly, M., Bryan, D., 1998. Hub location with flow economies of scale. Transportation Research Part B: Methodological 32, 605–616.
- O'Kelly, M.E., 1986a. The Location of Interacting Hub Facilities. Transportation Science 20, 92–106.
- O'Kelly, M.E., 1986b. Activity Levels at Hub Facilities in Interacting Networks. Geographical Analysis 18, 343–356.
- Özener, O.Ö., Ergun, Ö., 2008. Allocating Costs in a Collaborative Transportation Procurement Network. Transportation Science 42, 146–165.
- Phelps, S., Köksalan, M., 2003. An Interactive Evolutionary Metaheuristic for Multiobjective Combinatorial Optimization. Management Science 49, 1726–1738.
- Racunica, I., Wynter, L., 2005. Optimal location of intermodal freight hubs. Transportation Research Part B: Methodological 39, 453–477.
- Rodrigue, J.-P., Debrie, J., Fremont, A., Gouvernal, E., 2010. Functions and actors of inland ports: European and North American dynamics. Journal of Transport Geography 18, 519–529.
- Rodrigue, J.-P., Guan, C., 2009. Port-hinterland divergence along the North-American eastern seaboard, in: Ports in Proximity: Competition and Coordination among Adjacent Seaports. Ashgate, Suurey, pp. 131–150.
- Rodrigue, J.-P., Notteboom, T., 2009. The terminalization of supply chains: reassessing the role of terminals in port/hinterland logistical relationships. Maritime Policy & Management 36, 165– 183.
- Rosenthal, R.W., 1973. A class of games possessing pure-strategy Nash equilibria. Int J Game Theory 2, 65–67.
- Roso, V., Woxenius, J., Lumsden, K., 2009. The dry port concept: connecting container seaports with the hinterland. Journal of Transport Geography 17, 338–345.

- Sirikijpanichkul, A., Van Dam, K.H., Ferreira, L., Lukszo, Z., 2007. Optimizing the Location of Intermodal Freight Hubs: An Overview of the Agent Based Modelling Approach. Journal of Transportation Systems Engineering and Information Technology 7, 71–81.
- Skyrms, B., 2001. The Stag Hunt. Proceedings and Addresses of the American Philosophical Association 75, 31.
- Sörensen, K., Vanovermeire, C., 2013. Bi-objective optimization of the intermodal terminal location problem as a policy-support tool. Computers in Industry 64, 128–135.
- UN-ESCAP (United Nations Economic and Social Comission for Asia and Pacific), 2005. Regional Shipping and Port Development Strategies (Container Traffic Forecast) (No. Report ST/ESCAP/2398). United Nations, New York.
- Van Der Horst, M.R., De Langen, P.W., 2008. Coordination in Hinterland Transport Chains: A Major Challenge for the Seaport Community. Maritime Econ Logistics 10, 108–129.
- Veenstra, A., Zuidwijk, R., van Asperen, E., 2012. The extended gate concept for container terminals: Expanding the notion of dry ports. Maritime Econ Logistics 14, 14–32.
- Wardrop, J.G., 1952. Road paper: Some theoretical aspects of road traffic research. ICE Proceedings: Engineering Divisions 1, 325–362.
- Yamada, T., Russ, B.F., Castro, J., Taniguchi, E., 2009. Designing Multimodal Freight Transport Networks: A Heuristic Approach and Applications. Transportation Science 43, 129–143.
- Zhang, M., Wiegmans, B., Tavasszy, L., 2013. Optimization of multimodal networks including environmental costs: A model and findings for transport policy. Computers in Industry 64, 136–145.
- Zuidwijk, R.A., Veenstra, A.W., 2014. The Value of Information in Container Transport. Transportation Science.

## Appendix A

Notation	metric	description
N	$N \in \mathbb{N}^*$	number of destinations
М	$M \in \mathbb{N}^*$	number of candidate locations for the extended gate
n <sub>j</sub>	$n_j \in \mathbb{N}^*$	number of containers requiring shipment to destination j
j	$j \in \left\{1;; N\right\}$	index for destinations
i	$i \in \{1;;M\}$	index for extended gates (i=0 is used for the deep-sea port)
$\delta_{0,i}$	km	distance from the port to extended gate i
$Z_{i,j}^1$	€/container	truck transportation cost from i to destination j
K	$K \in \mathbb{R}_{+}^{*}$	number of container shipped by train
$Z_{0,i}^2(K)$	€/container	train transportation cost from the port to extended gate i
$Z_{i}^{1,2}(K)$	€/container	transshipment cost at the extended gate i
$Z_{0,i}^{3}(K)$	€/container	train + transshipment costs from the port to extended gate i
$X_{i,j}$	$0 \le X_{i,j} \le 1$	portion of the demand at destination j shipped via i
X <sub>i</sub>	$X_i = (X_{i,j})_{j \in M}$	strategy vector
y <sub>i</sub>	binary	<i>y</i> <sub>i</sub> =1 <i>if extended gate i is opened</i>
$P_{i,j}(X_{i,j},K)$	$P_{i,i}(X_{i,i},K) \in \mathbb{R}$	cost savings of a strategy compared to using only direct shipment
$L_i$		number of UE associated to the allocation game with extended gate i
${U}_{i,l}$	$\overline{U}_{i,l} \subset \{1;;N\}$	set of destinations using the extended gate under equilibrium l
$K^{i,j}$	$K_{i,j} \in \mathbb{N}^*$	minimum volume for intermodal transportation for destination j
$U_{i,\underline{l}}^{comp}$	$U_{i,l}^{comp} \subset P(U_{i,l})$	Set of sets compatible with user equilibrium l
$Eff(U_{i,l})$	$Eff(U_{i,l}) \in \mathbb{N}^*$	number of destinations that need to be convince to reach the equilibrium
$PoS_i$	$\overline{PoS_i \in \mathbb{R}_+}$	price of stability for location i
$S_i$	$S_i \subset \{1;; N\}$	set of destinations using extended gate i under the SO allocation

### **Appendix B**

### **Proof of Theorem 1**

### UE:

Let  $X_i = (X_{i,j})_{j \in N}$  be a user equilibrium of the allocation game (if any) and let  $K = \sum_{j \in N} n_j X_{i,j}$ . By contradiction, assume that that  $1 > X_{i,j} > 0$  for a given  $j \in \{1,...,N\}$ . This implies that  $Z_{i,j}(X_{i,j},K) < n_j Z_{0,j}^1$  with  $Z_{i,j}(X_{i,j},K)$  defined in Formula (10). Thus  $n_j(Z_{i,j}^1 + Z_{0,i}^3(K)) < n_j Z_{0,j}^1$ . Because  $Z_{0,i}^3$  is non-increasing in K, we obtain that  $n_j(Z_{i,j}^1 + Z_{0,j}^3(K + (1 - X_{i,j})n_j)) < n_j Z_{0,j}^1$ . This contradicts the fact that  $X_i$  is a user equilibrium as choosing  $X_{i,j} = 1$  enables reducing  $Z_{i,j}(X_{i,j},K)$ . **SO allocation:** 

Let  $X_i = (X_{i,j})_{j \in M}$  be a system optimum allocation (if any). By contradiction, assume that that  $1 > X_{i,j} > 0$  for a given  $j \in \{1, ..., N\}$ . Using the results obtained for UE, we can conclude that choosing  $X_{i,j} = 1$  will reduce the transportation cost for destination *j* without increasing the costs for the other destinations (as this implies increasing *K*, and  $Z_{0,i}^3$  is non-increasing in *K*). We consequently end up with a contradiction. This concludes the proof.

### **Proof of Theorem 2**

We can construct a UE as follows: We begin by setting  $X_{i,j} = 0 \quad \forall j \in \{1, ..., N\}$ . If none of the players can increase their profit by individually using the extended gate, then considering direct shipment for all the players is a UE. Otherwise, there is a destination  $k \in \{1, ..., N\}$  such that  $Z_{i,k}(X_{i,k}, K \mid X_{i,k} = 1) < Z_{i,k}(X_{i,k}, K \mid X_{i,k} = 0)$ . Let  $X_{i,k} = 1$ . If none of the remaining destinations can increase their profit by individually joining the extended gate, then considering direct shipment for all the players except for player k is a UE. Otherwise, the same procedure can be repeated as a destination included in the set of players using the extended gate at one step will never have any incentive to change its decision to direct shipment (because  $Z_{0,i}^3$  is non-increasing in K). Because  $\{1, ..., N\}$  is a finite set, the proposed procedure necessarily converges; thus, a UE always exists for the shippers' allocation game. This concludes the proof.

### **Proof of Theorem 3**

Consider an instance of the 2 shippers' allocation game with the following characteristics:  $n_1 = 2$ ,  $n_2 = 2$ ,  $K^{i,1} = 4$  and  $K^{i,2} = 4$ . We can notice that  $P_{i,1}(X_{i,1} = 1; X_{i,2} = 0) < 0$  and  $P_{i,2}(X_{i,1} = 0; X_{i,2} = 1) < 0$ . We conclude that  $X_i = (0;0)$  is a user equilibrium allocation for the game. Moreover,  $P_{i,1}(X_{i,1} = 1; X_{i,2} = 1) > 0$  and  $P_{i,2}(X_{i,1} = 1; X_{i,2} = 1) > 0$  as  $K^{i,1} = K^{i,2} \ge n_1 + n_2$ , thus,  $X_i = (1;1)$  is a second user equilibrium allocation for the game. This concludes the proof. We refer to Appendix C and Table 2 for a more detailed example with several UE.

### **Proof of Theorem 4**

Assume that there exist two destinations  $a, b \in \{1, ..., N\}$  such that  $a \neq b$  and  $K^{i,b} \geq K^{i,a}$ . Let  $U_{i,l}$  be a user equilibrium of the shippers' allocation game  $(l \in \{1, ..., L_i\})$  and let  $K_l = \sum_{j \in U_{i,l}} n_j$ . By definition of

$$K^{i,j}, b \in U_{i,l} \iff K^{i,b} \le K_l \text{ and } a \in U_{i,l} \iff K^{i,a} \le K_l.$$
 Then:

 $b \in U_{i,l} \implies K^{i,b} \le K_l \implies K^{i,a} \le K_l \implies a \in U_{i,l}$ . This concludes the proof.

### **Proof of Corollary 1**

Consider an instance of the shippers' allocation game with  $L_i \ge 2$  user equilibria. Let  $U_{i,11}$  and  $U_{i,12}$  be two distinct user equilibria. If  $U_{i,11} = \emptyset$  then  $U_{i,11} \subset U_{i,12}$  and if  $U_{i,12} = \emptyset$  then  $U_{i,12} \subset U_{i,11}$ . Let  $a \in \{1, ...; N\}$  such that  $a \in U_{i,11}$  and  $K^{i,a} = \max_{j \in U_{i,11}} (K^{i,j})$  and let  $b \in \{1, ...; N\}$  such that  $b \in U_{i,12}$  and  $K^{i,b} = \max_{j \in U_{i,12}} (K^{i,j})$ . If  $K^{i,a} \ge K^{i,b}$  then all the destinations included in  $U_{i,11}$  are included in  $U_{i,12}$  by applying Theorem 4 so  $U_{i,12} \subset U_{i,11}$ . Else,  $K^{i,a} < K^{i,b}$  then all destinations included in  $U_{i,12}$  are included in  $U_{i,12}$  are included in  $U_{i,12}$ . Else,  $K^{i,a} < K^{i,b}$  then all destinations included in  $U_{i,12}$  are included in  $U_{i,12}$ . Such that  $b \in U_{i,12}$  are included in  $U_{i,12}$ . This concludes the proof.

### **Proof of Corollary 2**

Consider an instance of the shippers' allocation game with  $L_i \ge 2$  user equilibria. By applying Corollary 1, we can deduce that  $Card(U_{i,l+1}) > Card(U_{i,l}), \forall l \in \{1, ..., L_i - 1\}$ . We aim at proving that  $\forall l \in \{1, ..., L_i - 1\}$ ,  $Card(U_{l+1}) \ge Card(U_l) + 2$ . By contradiction, assume that there is  $l \in \{1, ..., L_i - 1\}$ , such that  $Card(U_{i,l+1}) = Card(U_{i,l}) + 1$ . Then, a single player is added to the set of players using the extended gate by applying Theorem 4. This player's profit is strictly greater than zero when using the extended given the other players' decisions; thus,  $U_{l+1}$  is not a user equilibrium. We conclude that  $Card(U_{l+1}) \ge Card(U_l) + 2$ . This leads to  $L_i \le \left\lfloor \frac{N}{2} \right\rfloor + 1$ . This concludes the proof.

### **Proof of Corollary 3**

Consider an instance of the shippers' allocation game with  $L_i \ge 2$  user equilibria. Let  $K_l = \sum_{j \in U_{i,l}} n_j$ for all  $l \in \{1,...,L_i\}$ . By applying Corollary 1, we can deduce that  $K_l < K_{l+1}$  for all  $l \in \{1,...,L_i-1\}$ . Thus,  $U_{i,l+1}$  performs strictly better than  $U_{i,l}$  in terms of extended gate utilization.

The total cost for all destinations  $j \in \{1;...;N\} \setminus U_{i,l+1}$  does not differ for  $U_{i,l}$  and  $U_{i,l+1}$ . The total cost for all destinations  $j \in U_{i,l}$  is less or equal for  $U_{i,l+1}$  as compared to  $U_{i,l}$  (Because  $Z_{0,i}^3$  is non-increasing in K). The total cost for all destinations  $j \in U_{i,l+1} \setminus U_{i,1}$  is strictly less for  $U_{i,l+1}$  as compared to  $U_{i,l}$  (as destination decide to use the extended gate only if this strictly decrease its cost compared to direct shipment). As  $U_{i,l+1} \setminus U_{i,1}$  is non-empty, we conclude that  $U_{i,l+1}$  performs strictly better than  $U_{i,l}$  in terms of total cost. This concludes the proof.

### **Proof of Theorem 5**

Let  $l \in \{1, ..., L_i\}$  and  $U_{i,l}$  be a UE of the shippers' allocation game.

 $U_{i,l}^{\min}$  is non empty as  $U_{i,l} \in U_{i,l}^{\min}$ . Let  $U \in U_{i,l}^{\min}$ . Then,  $U \subseteq U_{i,l}$  and  $\forall k \in U$ ,  $K^{i,k} \leq \sum_{j \in U} n_j$ .

If  $U_{i,l} \setminus U = \emptyset$ , then  $U = U_{i,l}$  and we conclude that if l > 1,  $U \not\subset U_{i,l-1}$  by applying Corollary 1.

If  $U_{il} \setminus U \neq \emptyset$ , consider U' with  $U \subseteq U' \subset U_{il}$ .

Let  $k = \min(k' \in (1,...,Card(U_{i,l}) - Card(U))|k' \notin U')$ . Such a k exists as  $U' \subset U_{i,l}$ . Then  $K^{i,k} \leq \sum_{j \in U} n_j + \sum_{j=1}^k n_j \leq \sum_{j \in U'} n_j + n_k$  as  $j \in U'$  for all  $j \in \{1,...;k-1\}$ . Hence U' is not a UE. We conclude that  $U = U^{COMP}$ 

conclude that  $U \in U_{i,l}^{comp}$ .

 $U_{i,l}^{comp} \text{ is non empty as } U_{i,l} \in U_{i,l}^{comp}. \text{ Let } U \in U_{i,l}^{comp}. \text{ So } U \subseteq U_{i,l}. K^{i,k} \leq \sum_{j \in U} n_j \text{ for all } k \in U. \text{ If } l > 1,$  $U \not\subset U_{i,l-1}. \text{ We will construct sequence } (1, \dots, Card(U_{i,l}) - Card(U)) \text{ such that for all } j \in (1, \dots, Card(U_{i,l}) - Card(U)): U_j = U \cup (1, \dots, j) \in U_{i,l}^{comp}.$ 

Let  $U_0 = U$ . Obviously,  $U_0 \in U_{i,l}^{comp}$ . Let  $j \in \{1, ...; Card(U_{i,l}) - Card(U) - 1\}$  and assume that  $U_j \in U_{i,l}^{comp}$ . Hence,  $K^{i,k} \leq \sum_{j \in U_j} n_j$  for all  $k \in U_j$ ,  $U_j \not\subset U_{i,l-1}$  and  $U_j \subset U_{i,l}$ . As  $U_j$  is not a UE,

there exists  $j+1 \in N \setminus U_j$  such that:  $K^{i,j+1} \leq \sum_{j \in U_j} n_j + n_{j+1}$ . Then,  $K^{i,j+1} \leq \sum_{j \in U_{i,l}} n_j + n_{j+1}$  and thus

 $j+1 \in U_{i,l}$  as otherwise, this would contradicts that  $U_{i,l}$  is a UE. We conclude that  $U_{j+1} \in U_{i,l}^{comp}$  and by construction, we conclude that  $U \in U_{i,l}^{\min}$ . This concludes the proof.

### **Proof of Theorem 6**

Let  $b \in S_i$ , and let  $a \in N$  such that  $Z_{0,a}^1 - Z_{i,a}^1 \ge Z_{0,b}^1 - Z_{i,b}^1$ . As  $b \in S_i$ , the last container of destination b increases the profit when being shipped via the extended gate, i.e.:

$$\sum_{j \in S_i} n_j Z_{0,i}^3 \left( \sum_{j \in S_i} n_j \right) - (Z_{0,b}^1 - Z_{i,b}^1) - \left( \sum_{j \in S_i} n_j - 1 \right) Z_{0,i}^3 \left( \sum_{j \in S_i} n_j - 1 \right) < 0 \Rightarrow Z_{0,b}^1 - Z_{i,b}^1 > \sum_{j \in S_i} n_j Z_{0,i}^3 \left( \sum_{j \in S_i} n_j - 1 \right) Z_{0,i}^3 \left( \sum_{j \in S_i}$$

As  $Z_{0,a}^1 - Z_{i,a}^1 \ge Z_{0,b}^1 - Z_{i,b}^1$ , we obtain that:

 $Z_{0,a}^{1} - Z_{i,a}^{1} > \sum_{j \in S_{i}} n_{j} Z_{0,i}^{3} \left( \sum_{j \in S_{i}} n_{j} \right) - \left( \sum_{j \in S_{i}} n_{j} - 1 \right) Z_{0,i}^{3} \left( \sum_{j \in S_{i}} n_{j} - 1 \right)$  meaning that at least one container of destinction  $n_{i}$  exceeded acts. By explaining

destination *a* enables increasing the profit when being shipped via the extended gate. By applying Theorem 1, we conclude that  $a \in S_i$ . This concludes the proof.

### **Proof of Lemma 1**

Let  $a, b \in \{1, ..., N\}$  such that  $Z_{0,a}^1 - Z_{i,a}^1 \ge Z_{0,b}^1 - Z_{i,b}^1$ . If  $K^{i,a} = 1$  then  $K^{i,a} \le K^{i,b}$ . Else,  $P_{i,a}(1, K^{i,a} - 1) < 0$  by definition of  $K^{i,a}$ .  $P_{i,a}(1, K^{i,a} - 1) < 0 \Rightarrow n_a \Big( Z_{0,a}^1 - Z_{i,a}^1 - Z_{0,i}^3 (K^{i,a} - 1) \Big) < 0 \Rightarrow Z_{0,a}^1 - Z_{i,a}^1 - Z_{0,i}^3 (K^{i,a} - 1) < 0 \Rightarrow$  $Z_{0,b}^1 - Z_{i,b}^1 - Z_{0,i}^3 (K^{i,a} - 1) < 0 \Rightarrow P_{i,b}(1; K^{i,a} - 1) < 0 \Rightarrow K^{i,b} > K^{i,a} - 1 \Rightarrow K^{i,b} \ge K^{i,a}$ . This concludes

the proof.

### **Proof of Lemma 2**

Any destination that is individually better when using the extended gate contributes to the global cost minimization because this decision has no negative effect on the other destinations. We conclude that  $U_{i,L_i} \subseteq S_i$ . This concludes the proof.

### Appendix C

### Data related to the example proposed in section 5

In this example, N = 25 and M = 10. The locations of the destinations and the possible extended gate are displayed in Figure 1C. These ones have been randomly generated. The crosses represent the destinations, the dots represent the possible extended gate location and the square represents the location of the port. The extended gate location numbers are also included in figure 1.



Figure 1C: Location of the destinations and possible extended gate location

The truck transportation costs are assumed to be linear in the distance traveled for each destination. The distance is calculated by considering the Euclidean norm. Table 1C provides the demand and location for each destination, as well as the truck transportation cost per container for direct shipment.

j	x <sub>j</sub>	<b>y</b> j	n <sub>j</sub>	Z <sup>1</sup> <sub>0,j</sub>	j	x <sub>j</sub>	Yj	n <sub>j</sub>	Z <sup>1</sup> <sub>0,j</sub>
1	180	231	1	236	14	18	276	2	186
2	216	248	1	454	15	297	235	1	332
3	265	21	2	494	16	48	231	2	106
4	241	81	2	330	17	259	146	1	265
5	196	25	2	318	18	260	3	1	470
6	55	141	1	67	19	26	252	2	146
7	30	186	6	64	20	114	84	1	233
8	193	187	2	380	21	196	185	1	248
9	236	50	1	277	22	54	97	3	138
10	234	224	2	451	23	96	172	1	175
11	135	292	2	254	24	189	97	3	366
12	261	32	1	455	25	159	205	3	187
13	106	103	2	138					

Table 1C: Demand and direct shipment cost for each destination

Transloading costs at the extended gates are assumed to be affine in function of the number of container transloaded, i.e., that  $KZ_i^{1,2}(K) = A_i + B_i K$ . The parameters appear in Table 2C.

i	Ai	Bi
1	74	11
2	91	8
3	80	3
4	2	10
5	0	7
6	11	4
7	44	12
8	50	4
9	43	1
10	39	3

Table 2C: Parameters for the calculation of the transloading cost

The train transportation cost is linear in the distance traveled. The total train transportation cost per kilometer is piecewise linear in the number of container transported and is calculated as follows,  $KZ_{0,i}^2(K) = a + bK + c.\min(d; K)$  with a = 8, b = 0,1, c = 0,3 and d = 15. The evaluation of  $K^{i,j}$  for all  $i \in \{1,...,M\}$  and for all  $j \in \{1,...,N\}$  can then be performed by calculating  $P_{i,j}(1; K)$  for all  $K \leq \sum_{j \in N} n_j$ . Table 3C provides some details of the calculations for extended gate 1 and destination 1.

We can deduce from Table 3C that  $K^{1,1} = 33$ .

К	20	21	22	23	24	25	26	27	28	29	30	31	32	33	34
Z <sup>1</sup> <sub>0,1</sub> -Z <sup>1</sup> <sub>1,1</sub>	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8	107,8
Z <sup>3</sup> <sub>0,1</sub> (K)	157,5	151,5	146,0	141,0	136,4	132,2	128,3	124,6	121,3	118,2	115,3	112,5	110,0	107,6	105,3
$P_{1,1}(1;K)$	-49,7	-43,7	-38,2	-33,2	-28,6	-24,4	-20,5	-16,9	-13,5	-10,4	-7,5	-4,7	-2,2	0,2	2,5

Table 3C: Details for the calculation of K<sup>1,1</sup>

Table 4C provides the value of  $K^{i,j}$  for all  $i \in \{1,...,M\}$  and for all  $j \in \{1,...,N\}$ .

j\ <sup>i</sup>	1	2	3	4	5	6	7	8	9	10
1	33	12	8	8	8	33	8	18	12	31
2	14	6	8	8	46	12	19	8	6	11
3	9	20	7	8	7	12	11	11	32	16
4	11	21	16	14	11	15	17	13	28	19
5	17	8	17	10	10	31	17	22	8	46
6	8	8	8	8	8	8	22	8	8	8
7	8	∞	8	8	8	8	8	8	8	8
8	10	7	8	27	25	12	12	6	8	13
9	17	36	17	16	13	23	24	20	8	29
10	10	7	8	39	25	9	16	7	7	8
11	8	20	8	8	8	8	8	44	11	8
12	9	21	8	9	7	13	12	12	32	17
13	8	8	8	20	8	8	20	8	8	8
14	8	8	8	8	8	8	8	8	8	8
15	19	14	8	8	33	16	41	14	14	14
16	8	8	8	8	8	8	8	8	8	8
17	16	19	32	29	22	14	30	15	22	16
18	11	26	8	8	8	16	12	14	45	19
19	8	8	8	8	8	8	8	8	8	8
20	8	8	8	8	37	8	10	8	8	8
21	18	13	8	8	42	20	25	11	15	21
22	8	8	8	33	8	8	12	8	8	8
23	8	∞	8	8	8	8	17	8	37	8
24	8	18	22	9	9	16	9	9	24	21
25	44	14	8	∞	∞	∞	∞	22	15	8

Table 4C: Values of K<sup>i,j</sup>

These data and the results provided in Section 4 enable us to identify all the existing UE as well as the SO allocation for all  $i \in \{1, ..., M\}$ . The results appear in Section 5.

Nr.	Year	Title	Author(s)
481	2015	The extended gate problem: Intermodal hub location with multiple actors	Yann Bouchery, Jan Fransoo, Marco Slikker
480	2015	Inventory Management with Two Demand Streams: A Maintenance Application	Rob J.I. Basten, Jennifer K. Ryan
479	2015	Optimal Design of Uptime-Guarantee Contracts	Behzad Hezarkhani
478	2015	Collaborative Replenishment in the Presence of Intermediaries	Behzad Hezarkhani, Marco Slikker, Tom Van Woensel
477	2015	Reference Architecture for Mobility-Related Services A reference architecture based on GET Service and SIMPLI-CITY Project architectures	A. Husak, M. Politis, V. Shah, R. Eshuis, P. Grefen
476	2015	A Multi-Item Approach to Repairable Stocking and Expediting in a Fluctuating Demand Environment	Joachim Arts
475	2015	An Adaptive Large Neighborhood Search Heuristic for the Share-a-Ride Problem	Baoxiang Li, Dmitry Krushinsky, Tom Van Woensel, Hajo A. Reijers
474	2015	An approximate dynamic programming approach to urban freight distribution with batch arrivals	Wouter van Heeswijk, Martijn Mes, Marco Schutten
473	2015	Dynamic Multi-period Freight Consolidation	Arturo Pérez Rivera, Martijn Mes
472	2015	Maintenance policy selection for ships: finding the most important criteria and considerations	A.J.M. Goossens, R.J.I. Basten
471	2015	Using Twitter to Predict Sales: A Case Study	Remco Dijkman, Panagiotis Ipeirotis, Freek Aertsen, Roy van Helden
470	2015	The Effect of Exceptions in Business Processes	Remco Dijkman, Geoffrey van IJzendoorn, Oktay Türetken, Meint de Vries
469	2015	Business Model Prototyping for Intelligent Transport Systems. A Service-Dominant Approach	Konstantinos Traganos, Paul Grefen, Aafke den Hollander, Oktay Türetken, Rik Eshuis
468	2015	How suitable is the RePro technique for rethinking care processes?	Rob J.B. Vanwersch, Luise Pufahl, Irene Vanderfeesten, Jan Mendling, Hajo A. Reijers
467	2014	Where to exert abatement effort for sustainable operations considering supply chain interactions?	Tarkan Tan, Astrid Koomen
466	2014	An Exact Algorithm for the Vehicle Routing Problem with Time Windows and Shifts	Said Dabia, Stefan Ropke, Tom Van Woensel
465	2014	The RePro technique: a new, systematic technique for rethinking care processes	Rob J.B. Vanwersch, Luise Pufahl, Irene Vanderfeesten, Hajo A. Reijers
464	2014	Exploring maintenance policy selection using the Analytic Hierarchy Process: an application for naval ships	A.J.M. Goossens, R.J.I. Basten
463	2014	Allocating service parts in two-echelon networks at a utility company	D. van den Berg, M.C. van der Heijden, P.C. Schuur
462	2014	Freight consolidation in networks with transshipments	W.J.A. van Heeswijk, M.R.K. Mes, J.M.J. Schutten, W.H.M. Zijm
461	2014	A Software Architecture for a Transportation Control Tower	Anne Baumgrass, Remco Dijkman, Paul Grefen, Shaya Pourmirza, Hagen Völzer, Mathias Weske
460	2014	Small traditional retailers in emerging markets	Youssef Boulaksil, Jan C. Fransoo, Edgar E. Blanco, Sallem Koubida
459	2014	Defining line replaceable units	J.E. Parada Puig, R.J.I. Basten
458	2014	Inventories and the Credit Crisis: A Chicken and Egg Situation	Maximiliano Udenio, Vishal Gaur, Jan C. Fransoo
457	2014	An Exact Approach for the Pollution-Routing Problem	Said Dabia, Emrah Demir, Tom Van Woensel
456	2014	Fleet readiness: stocking spare parts and high-tech assets	Rob J.I. Basten, Joachim J. Arts

Nr.	Year	Title	Author(s)
455	2014	Competitive Solutions for Cooperating Logistics Providers	Behzad Hezarkhani, Marco Slikker, Tom Van Woensel
454	2014	Simulation Framework to Analyse Operating Room Release Mechanisms	Rimmert van der Kooij, Martijn Mes, Erwin Hans
453	2014	A Unified Race Algorithm for Offline Parameter Tuning	Tim van Dijk, Martijn Mes, Marco Schutten, Joaquim Gromicho
452	2014	Cost, carbon emissions and modal shift in intermodal network design decisions	Yann Bouchery, Jan Fransoo
451	2014	Transportation Cost and CO2 Emissions in Location Decision Models	Josue C. Vélazquez-Martínez, Jan C. Fransoo, Edgar E. Blanco, Jaime Mora- Vargas
450	2014	Tracebook: A Dynamic Checklist Support System	Shan Nan, Pieter Van Gorp, Hendrikus H.M. Korsten, Richard Vdovjak, Uzay Kaymak
449	2014	Intermodal hinterland network design with multiple actors	Yann Bouchery, Jan Fransoo
448	2014	The Share-a-Ride Problem: People and Parcels Sharing Taxis	Baoxiang Li, Dmitry Krushinsky, Hajo A. Reijers, Tom Van Woensel
447	2014	Stochastic inventory models for a single item at a single location	K.H. van Donselaar, R.A.C.M. Broekmeulen
446	2014	Optimal and heuristic repairable stocking and expediting in a fluctuating demand environment	Joachim Arts, Rob Basten, Geert-Jan van Houtum
445	2014	Connecting inventory control and repair shop control: a differentiated control structure for repairable spare parts	M.A. Driessen, W.D. Rustenburg, G.J. van Houtum, V.C.S. Wiers
444	2014	A survey on design and usage of Software Reference Architectures	Samuil Angelov, Jos Trienekens, Rob Kusters
443	2014	Extending and Adapting the Architecture Tradeoff Analysis Method for the Evaluation of Software Reference Architectures	Samuil Angelov, Jos J.M. Trienekens, Paul Grefen
442	2014	A multimodal network flow problem with product quality preservation, transshipment, and asset management	Maryam SteadieSeifi, Nico Dellaert, Tom Van Woensel
441	2013	Integrating passenger and freight transportation: Model formulation and insights	Veaceslav Ghilas, Emrah Demir, Tom Van Woensel
440	2013	The Price of Payment Delay	K. van der Vliet, M.J. Reindorp, J.C.
439	2013	On Characterization of the Core of Lane Covering Games via Dual Solutions	Behzad Hezarkhani, Marco Slikker, Tom van Woensel
438	2013	Destocking, the Bullwhip Effect, and the Credit Crisis: Empirical Modeling of Supply Chain Dynamics	Maximiliano Udenio, Jan C. Fransoo, Robert Peels
437	2013	Methodological support for business process redesign in healthcare: a systematic literature review	Rob J.B. Vanwersch, Khurram Shahzad, Irene Vanderfeesten, Kris Vanhaecht, Paul Grefen, Liliane Pintelon, Jan Mendling, Geofridus G. van Merode, Hajo A. Reijers
436	2013	Dynamics and equilibria under incremental horizontal differentiation on the Salop circle	B. Vermeulen, J.A. La Poutré, A.G. de Kok
435	2013	Analyzing Conformance to Clinical Protocols Involving Advanced Synchronizations	Hui Yan, Pieter Van Gorp, Uzay Kaymak, Xudong Lu, Richard Vdovjak, Hendriks H.M. Korsten, Huilong Duan
434	2013	Models for Ambulance Planning on the Strategic and the Tactical Level	J. Theresia van Essen, Johann L. Hurink, Stefan Nickel, Melanie Reuter
433	2013	Mode Allocation and Scheduling of Inland Container Transportation: A Case-Study in the Netherlands	Stefano Fazi, Tom Van Woensel, Jan C. Fransoo
432	2013	Socially responsible transportation and lot sizing: Insights from multiobjective optimization	Yann Bouchery, Asma Ghaffari, Zied Jemai, Jan Fransoo
431	2013	Inventory routing for dynamic waste collection	Martijn Mes, Marco Schutten, Arturo Pérez Rivera

Nr.	Year	Title	Author(s)
430	2013	Simulation and Logistics Optimization of an Integrated Emergency Post	N.J. Borgman, M.R.K. Mes, I.M.H. Vliegen, E.W. Hans
429	2013	Last Time Buy and Repair Decisions for Spare Parts	S. Behfard, M.C. van der Heijden, A. Al Hanbali, W.H.M. Ziim
428	2013	A Review of Recent Research on Green Road Freight Transportation	Emrah Demir, Tolga Bektas, Gilbert Laporte
427	2013	Typology of Repair Shops for Maintenance Spare Parts	M.A. Driessen, V.C.S. Wiers, G.J. van
426	2013	A value network development model and implications for innovation and production network management	B. Vermeulen, A.G. de Kok
425	2013	Single Vehicle Routing with Stochastic Demands:	C. Zhang, N.P. Dellaert, L. Zhao, T. Van Woensel, D. Sever
424	2013	Influence of Spillback Effect on Dynamic Shortest Path Problems with Travel-Time-Dependent Network Disruptions	Derya Sever, Nico Dellaert, Tom Van Woensel, Ton de Kok
423	2013	Dynamic Shortest Path Problem with Travel-Time- Dependent Stochastic Disruptions: Hybrid Approximate Dynamic Programming Algorithms with a Clustering Approach	Derya Sever, Lei Zhao, Nico Dellaert, Tom Van Woensel, Ton de Kok
422	2013	System-oriented inventory models for spare parts	R.J.I. Basten, G.J. van Houtum
421	2013	Lost Sales Inventory Models with Batch Ordering and Handling Costs	T. Van Woensel, N. Erkip, A. Curseu, J.C. Fransoo
420	2013	Response speed and the bullwhip	Maximiliano Udenio, Jan C. Fransoo, Eleni Vatamidou, Nico Dellaert
419	2013	Anticipatory Routing of Police Helicopters	Rick van Urk, Martijn R.K. Mes, Erwin W. Hans
418	2013	Supply Chain Finance: research challenges ahead	Kasper van der Vliet, Matthew J. Reindorp, Jan C. Fransoo
417	2013	Improving the Performance of Sorter Systems by Scheduling Inbound Containers	S.W.A. Haneyah, J.M.J. Schutten, K. Fikse
416	2013	Regional logistics land allocation policies: Stimulating spatial concentration of logistics firms	Frank P. van den Heuvel, Peter W. de Langen, Karel H. van Donselaar, Jan C. Fransoo
415	2013	The development of measures of process harmonization	Heidi L. Romero, Remco M. Dijkman, Paul W.P.J. Grefen, Arjan van Weele
414	2013	BASE/X. Business Agility through Cross-Organizational Service Engineering. The Business and Service Design Approach developed in the CoProFind Project	Paul Grefen, Egon Lüftenegger, Eric van der Linden, Caren Weisleder
413	2013	The Time-Dependent Vehicle Routing Problem with Soft Time Windows and Stochastic Travel Times	Duygu Tas, Nico Dellaert, Tom van Woensel, Ton de Kok
412	2013	Clearing the Sky - Understanding SLA Elements in Cloud Computing	Marco Comuzzi, Guus Jacobs, Paul Grefen
411	2013	Approximations for the waiting time distribution in an M/G/c priority queue	A. Al Hanbali, E.M. Alvarez, M.C. van der Heijden
410	2013	To co-locate or not? Location decisions and logistics concentration areas	Frank P. van den Heuvel, Karel H. van Donselaar, Rob A.C.M. Broekmeulen, Jan C. Fransoo, Peter W. de Langen
409	2013	The Time-Dependent Pollution-Routing Problem	Anna Franceschetti, Dorothée Honhon, Tom van Woensel, Tolga Bektas, Gilbert Laporte
408	2013	Scheduling the scheduling task: A time management perspective on scheduling	J.A. Larco, V. Wiers, J. Fransoo
407	2013	Clustering Clinical Departments for Wards to Achieve a Prespecified Blocking Probability	J. Theresia van Essen, Mark van Houdenhoven, Johann L. Hurink

Nr.	Year	Title	Author(s)
406	2013	MyPHRMachines: Personal Health Desktops in the Cloud	Pieter Van Gorp, Marco Comuzzi
405	2013	Maximising the Value of Supply Chain Finance	Kasper van der Vliet, Matthew J. Reindorp, Jan C. Fransoo
404	2013	Reaching 50 million nanostores: retail distribution in emerging megacities	Edgar E. Blanco, Jan C. Fransoo
403	2013	A Vehicle Routing Problem with Flexible Time Windows	Duygu Tas, Ola Jabali, Tom van Woensel
402	2013	The Service Dominant Business Model: A Service Focused Conceptualization	Egon Lüftenegger, Marco Comuzzi, Paul Grefen, Caren Weisleder
401	2013	Relationship between freight accessibility and logistics employment in US counties	Frank P. van den Heuvel, Liliana Rivera, Karel H. van Donselaar, Ad de Jong, Yossi Sheffi, Peter W. de Langen, Jan C. Fransoo
400	2012	A Condition-Based Maintenance Policy for Multi-Component Systems with a High Maintenance Setup Cost	Qiushi Zhu, Hao Peng, Geert-Jan van Houtum
399	2012	A flexible iterative improvement heuristic to support creation of feasible shift rosters in self-rostering	E. van der Veen, J.L. Hurink, J.M.J. Schutten, S.T. Uijland
398	2012	Scheduled Service Network Design with Synchronization and Transshipment Constraints for Intermodal Container Transportation Networks	K. Sharypova, T.G. Crainic, T. van Woensel, J.C. Fransoo
397	2012	Destocking, the bullwhip effect, and the credit crisis: empirical modeling of supply chain dynamics	Maximiliano Udenio, Jan C. Fransoo, Robert Peels
396	2012	Vehicle routing with restricted loading capacities	J. Gromicho, J.J. van Hoorn, A.L. Kok, J.M.J. Schutten
395	2012	Service differentiation through selective lateral transshipments	E.M. Alvarez, M.C. van der Heijden, I.M.H. Vliegen, W.H.M. Zijm
394	2012	A Generalized Simulation Model of an Integrated Emergency Post	Martijn Mes, Manon Bruens
393	2012	Business Process Technology and the Cloud: defining a Business Process Cloud Platform	Vassil Stoitsev, Paul Grefen
392	2012	Vehicle Routing with Soft Time Windows and Stochastic Travel Times: A Column Generation and Branch-and-Price Solution Approach	D. Tas, M. Gendreau, N. Dellaert, T. van Woensel, A.G. de Kok
391	2012	Improve OR-Schedule to Reduce Number of Required Beds	J. Theresia van Essen, Joël M. Bosch, Erwin W. Hans, Mark van Houdenhoven, Johann L. Hurink
390	2012	How does development lead time affect performance over the ramp-up lifecycle? Evidence from the consumer electronics industry	Andreas Pufall, Jan C. Fransoo, Ad de Jong, A.G. (Ton) de Kok
389	2012	The Impact of Product Complexity on Ramp-Up Performance	Andreas Pufall, Jan C. Fransoo, Ad de Jong, A.G. (Ton) de Kok
388	2012	Co-location synergies: specialized versus diverse logistics concentration areas	Frank P. van den Heuvel, Peter W. de Langen, Karel H. van Donselaar, Jan C. Fransoo
387	2012	Proximity matters: Synergies through co-location of logistics establishments	Frank P. van den Heuvel, Peter W. de Langen, Karel H. van Donselaar, Jan C. Fransoo
386	2012	Spatial concentration and location dynamics in logistics: the case of a Dutch province	Frank P. van den Heuvel, Peter W. de Langen, Karel H. van Donselaar, Jan C. Fransoo
385	2012	FNet: An Index for Advanced Business Process Querying	Zhiqiang Yan, Remco Dijkman, Paul Grefen
384	2012	Defining Various Pathway Terms	W.R. Dalinghaus, P.M.E. Van Gorp
383	2012	The Service Dominant Strategy Canvas: Defining and Visualizing a Service Dominant Strategy through the Traditional Strategic Lens	Egon Lüftenegger, Paul Grefen, Caren Weisleder

Nr.	Year	Title	Author(s)
382	2012	A Stochastic Variable Size Bin Packing Problem with Time	Stefano Fazi, Tom van Woensel, Jan C.
		Constraints	Fransoo
381	2012	Coordination and Analysis of Barge Container Hinterland	K. Sharypova, T. van Woensel, J.C.
		Networks	Fransoo
380	2012	Proximity matters: Synergies through co-location of logistics	Frank P. van den Heuvel, Peter W. de
		establishments	Langen, Karel H. van Donselaar, Jan C.
			Fransoo
379	2012	A literature review in process harmonization: a conceptual	Heidi Romero, Remco Dijkman, Paul
		framework	Grefen, Arjan van Weele
378	2012	A Generic Material Flow Control Model for Two Different	S.W.A. Haneyah, J.M.J. Schutten, P.C.
		Industries	Schuur, W.H.M. Zijm
377	2012	Dynamic demand fulfillment in spare parts networks with	H.G.H. Tiemessen, M. Fleischmann, G.J.
		multiple customer classes	van Houtum, J.A.E.E. van Nunen, E.
			Pratsini
376	2012	Paper has been replaced by wp 417	K. Fikse, S.W.A. Haneyah, J.M.J.
	0.010		Schutten
375	2012	Strategies for dynamic appointment making by container	Albert Douma, Martijn Mes
274	2012	terminais	Distances Com Manag Comuni
374	2012	NyPHRMachines: Lifelong Personal Health Records in the	Pleter van Gorp, Marco Comuzzi
272	2012	Cloud	E M Alvaraz M C van dar Heijden
5/5	2012	dedicated stocks	W H M Ziim
272	2012	Spare parts inventory pooling: how to share the benefits?	Frank Karston, Rob Baston
572	2012	spare parts inventory pooling. now to share the benefits!	Frank Karsten, KOD Basten
371	2012	Condition based spare parts supply	X Lin R LL Basten A A Kranenburg
571			G.J. van Houtum
370	2012	Using Simulation to Assess the Opportunities of Dynamic	Martiin Mes
	-	Waste Collection	
369	2012	Aggregate overhaul and supply chain planning for rotables	J. Arts, S.D. Flapper, K. Vernooij
368	2012	Operating Room Rescheduling	J.T. van Essen, J.L. Hurink, W. Hartholt,
			B.J. van den Akker
367	2011	Switching Transport Modes to Meet Voluntary Carbon	Kristel M.R. Hoen, Tarkan Tan, Jan C.
		Emission Targets	Fransoo, Geert-Jan van Houtum
366	2011	On two-echelon inventory systems with Poisson demand	Elisa Alvarez, Matthieu van der Heijden
		and lost sales	
365	2011	Minimizing the Waiting Time for Emergency Surgery	J.T. van Essen, E.W. Hans, J.L. Hurink, A.
			Oversberg
364	2012	Vehicle Routing Problem with Stochastic Travel Times	Duygu Tas, Nico Dellaert, Tom van
		Including Soft Time Windows and Service Costs	Woensel, Ton de Kok
363	2011	A New Approximate Evaluation Method for Two-Echelon	Erhun Ozkan, Geert-Jan van Houtum,
		Inventory Systems with Emergency Shipments	Yasemin Serin
262	2011		
362	2011	Approximating Multi-Objective Time-Dependent	Said Dabia, El-Gnazali Taibi, Tom Van
261	2011	Optimization Problems	Woensel, Ton de Kok
201	2011	Branch and Cut and Price for the Time Dependent vehicle	Woonsol Ton do Kok
260	2011	Analysis of an Assemble to Order System with Different	A C Karaarslan C D Kinsmüller A C de
300	2011	Review Periods	Kok
359	2011	Interval Availability Analysis of a Two-Echelon, Multi-Item	Ahmad Al Hanbali, Matthieu van der
555	2011	System	Heiiden
358	2011	Carbon-Ontimal and Carbon-Neutral Supply Chains	Felipe Caro, Charles I, Corbett, Tarkan
			Tan, Rob Zuidwiik
357	2011	Generic Planning and Control of Automated Material	Sameh Haneyah, Henk Ziim. Marco
		Handling Systems: Practical Requirements Versus Existing	Schutten, Peter Schuur
		Theory	
356	2011	Last time buy decisions for products sold under warranty	Matthieu van der Heijden, Bermawi
			Iskandar

Nr.	Year	Title	Author(s)
355	2011	Spatial concentration and location dynamics in logistics: the	Frank P. van den Heuvel, Peter W. de
		case of a Dutch province	Langen, Karel H. van Donselaar, Jan C.
			Fransoo
354	2011	Identification of Employment Concentration Areas	Frank P. van den Heuvel, Peter W. de
			Langen, Karel H. van Donselaar, Jan C.
			Fransoo
353	2011	BPMN 2.0 Execution Semantics Formalized as Graph Rewrite	Pieter van Gorp, Remco Dijkman
		Rules: extended version	
352	2011	Resource pooling and cost allocation among independent	Frank Karsten, Marco Slikker, Geert-Jan
		service providers	van Houtum
351	2011	A Framework for Business Innovation Directions	E. Lüftenegger, S. Angelov, P. Grefen
350	2011	The Road to a Business Process Architecture: An Overview of	Remco Diikman, Irene Vanderfeesten
550	2011	Annroaches and their Lise	Haio A Reijers
3/19	2011	Effect of carbon emission regulations on transport mode	K M R Hoen T Tan I C Fransoo G I
545	2011	selection under stochastic demand	van Houtum
2/10	2011	An improved MIP-based combinatorial approach for a multi-	Murat Firat, Cor Hurkons
540	2011	skill workforce scheduling problem	
3/17	2011	An approximate approach for the joint problem of level of	R I I Basten M C van der Heijden
547	2011	repair approximate approach for the joint problem of level of	I.M.I. Schutton
346	2011	Ioint ontimization of level of renair analysis and share parts	R LL Basten M C van der Heijden
540	2011	stocks	I.M.I. Schutton
245	2011	Stocks	Ton G. do Kok
545	2011		
344	2011	Analysis of resource pooling games via a new extension of	Frank Karsten, Marco Slikker, Geert-Jan
		the Erlang loss function	van Houtum
343	2011	Vehicle refueling with limited resources	Murat Firat, C.A.J. Hurkens, Gerhard J.
			Woeginger
342	2011	Optimal Inventory Policies with Non-stationary Supply	Bilge Atasoy, Refik Güllü, Tarkan Tan
		Disruptions and Advance Supply Information	
341	2011	Redundancy Optimization for Critical Components in High-	Kurtulus Baris Öner, Alan Scheller-Wolf,
		Availability Capital Goods	Geert-Jan van Houtum
340	2011	Making Decision Process Knowledge Explicit Using the	Razvan Petrusel, Irene Vanderfeesten,
		Product Data Model	Cristina Claudia Dolean, Daniel Mican
339	2010	Analysis of a two-echelon inventory system with two supply	Joachim Arts, Gudrun Kiesmüller
		modes	
338	2010	Analysis of the dial-a-ride problem of Hunsaker and	Murat Firat, Gerhard J. Woeginger
		Savelsbergh	
335	2010	Attaining stability in multi-skill workforce scheduling	Murat Firat, Cor Hurkens
334	2010	Flexible Heuristics Miner (FHM)	A.J.M.M. Weijters, J.T.S. Ribeiro
333	2010	An exact approach for relating recovering surgical patient	P.T. Vanberkel, R.J. Boucherie, E.W.
		workload to the master surgical schedule	Hans, J.L. Hurink, W.A.M. van Lent, W.H.
			van Harten
332	2010	Efficiency evaluation for pooling resources in health care	Peter T. Vanberkel, Richard J.
			Boucherie, Erwin W. Hans, Johann L.
			Hurink, Nelly Litvak
331	2010	The Effect of Workload Constraints in Mathematical	M.M. Jansen, A.G. de Kok, I.J.B.F. Adan
		Programming Models for Production Planning	, , ,
330	2010	Using pipeline information in a multi-echelon spare parts	Christian Howard. Ingrid Reiinen. Johan
		inventory system	Marklund. Tarkan Tan
329	2010	Reducing costs of repairable spare parts supply systems via	H.G.H. Tiemessen. G.J. van Houtum
		dynamic scheduling	
328	2010	Identification of Employment Concentration and	Frank P. van den Heuvel, Peter W. de
		Specialization Areas: Theory and Application	Langen, Karel H. van Donselaar, Jan C
			Fransoo
327	2010	A combinatorial approach to multi-skill workforce scheduling	M Firat C Hurkens
527			

Nr.	Year	Title	Author(s)
326	2010	Stability in multi-skill workforce scheduling	M. Firat, C. Hurkens, A. Laugier
325	2010	Maintenance spare parts planning and control: A framework	M.A. Driessen, J.J. Arts. G.J. van
		for control and agenda for future research	Houtum, W.D. Rustenburg, B. Huisman
324	2010	Near-optimal heuristics to set base stock levels in a two- echelon distribution network	R.J.I. Basten, G.J. van Houtum
323	2010	Inventory reduction in spare part networks by selective	M.C. van der Heijden, E.M. Alvarez,
		throughput time reduction	J.M.J. Schutten
322	2010	The selective use of emergency shipments for service-	E.M. Alvarez, M.C. van der Heijden,
		contract differentiation	W.H.M. Zijm
321	2010	Heuristics for Multi-Item Two-Echelon Spare Parts Inventory Control Problem with Batch Ordering in the Central Warehouse	Engin Topan, Z. Pelin Bayindir, Tarkan Tan
320	2010	Preventing or escaping the suppression mechanism:	Bob Walrave, Kim F, van Oorschot, A
520	2010	intervention conditions	Georges L. Romme
319	2010	Hospital admission planning to optimize major resources	Nico Dellaert, Jully Jeunet
515	2010	utilization under uncertainty	
318	2010	Minimal Protocol Adaptors for Interacting Services	R. Seguel, R. Eshuis, P. Grefen
317	2010	Teaching Retail Operations in Business and Engineering	Tom Van Woensel, Marshall L. Fisher.
51/	2010	Schools	lan C Fransoo
316	2010	Design for Availability: Creating Value for Manufacturers and	I vdie P.M. Smets. Geert-Ian van
510	2010	Customers	Houtum Fred Langerak
315	2010	Transforming Process Models: executable rewrite rules	Pieter van Gorn Rik Eshuis
515	2010	versus a formalized lava program	
31/	2010	Working paper 314 is no longer available	
313	2010	A Dynamic Programming Approach to Multi-Objective Time-	S Dabia T van Woensel A G de Kok
515	2010	Dependent Canacitated Single Vehicle Routing Problems	
		with Time Windows	
312	2010	Tales of a Solu)rcerer: Ontimal Sourcing Decisions Linder	Osman Aln, Tarkan Tan
512	2010	Alternative Canacitated Suppliers and General Cost	
		Structures	
311	2010	In-store replenishment procedures for perishable inventory	RACM Broekmeulen CHM Baky
511	2010	in a retail environment with handling costs and storage	
		constraints	
310	2010	The state of the art of innovation-driven business models in	F Lüftenegger S Angelov F van der
510	2010	the financial services industry	Linden P Grefen
309	2010	Design of Complex Architectures Using a Three Dimension	R Seguel P Grefen R Eshuis
505	2010	Annroach: the CrossWork Case	R. Seguel, F. Greren, R. Eshuis
308	2010	Effect of carbon emission regulations on transport mode	KMR Hoen T Tan IC Fransoo G I
500	2010	selection in supply chains	van Houtum
307	2010	Interaction between intelligent agent strategies for real-time	Martiin Mes. Matthieu van der Heiiden
507	2010	transportation planning	Peter Schuur
306	2010	Internal Slackening Scoring Methods	Marco Slikker, Peter Borm, Bené van
500	2010		den Brink
305	2010	Vehicle Bouting with Traffic Congestion and Drivers' Driving	A L Kok F W Hans I M L Schutten
505	2010	and Working Rules	W H M 7iim
30/	2010	Practical extensions to the level of renair analysis	R I I Basten M C van der Heijden
504	2010		I M I Schutten
303	2010	Ocean Container Transport: An Underestimated and Critical	lan C Fransoo Chung-Vee Lee
505	2010	Link in Global Supply Chain Performance	
202	2010	Canacity recervation and utilization for a manufacturer with	V Boulaksil: I.C. Eransoo: T. Tan
502	2010	uncertain capacity and demand	
300	2000	Spare parts inventory pooling games	E I B. Karston: M. Slikker: G I van
300	2009	Spare parts inventory pooling games	Houtum
200	2009	Canacity flexibility allocation in an outcoursed supply chain	V Boulaksil M Grupow LC Frances
200	2003	with reservation	T. Doulaksii, M. Orunow, J.C. Fransoo
298	2010	An ontimal approach for the joint problem of level of repair	R I I Basten M C van der Heijden
2.50	2010	analysis and snare narts stocking	I M I Schutten
1	1		STATIST SCHULLEH

Nr.	Year	Title	Author(s)
297	2009	Responding to the Lehman Wave: Sales Forecasting and	Robert Peels, Maximiliano Udenio, Jan
		Supply Management during the Credit Crisis	C. Fransoo, Marcel Wolfs, Tom Hendrikx
296	2009	An exact approach for relating recovering surgical patient	Peter T. Vanberkel, Richard J.
		workload to the master surgical schedule	Boucherie, Erwin W. Hans, Johann L.
			Hurink, Wineke A.M. van Lent, Wim H.
			van Harten
295	2009	An iterative method for the simultaneous optimization of	R.J.I. Basten, M.C. van der Heijden,
		repair decisions and spare parts stocks	J.M.J. Schutten
294	2009	Fujaba hits the Wall(-e)	Pieter van Gorp, Ruben Jubeh, Bernhard
			Grusie, Anne Keller
293	2009	Implementation of a Healthcare Process in Four Different	R.S. Mans, W.M.P. van der Aalst, N.C.
		Workflow Systems	Russell, P.J.M. Bakker
292	2009	Business Process Model Repositories - Framework and	Zhiqiang Yan, Remco Dijkman, Paul
		Survey	Grefen
291	2009	Efficient Optimization of the Dual-Index Policy Using Markov	Joachim Arts, Marcel van Vuuren,
		Chains	Gudrun Kiesmuller
290	2009	Hierarchical Knowledge-Gradient for Sequential Sampling	Martijn R.K. Mes; Warren B. Powell;
	<u> </u>		Peter I. Frazier
289	2009	Analyzing combined vehicle routing and break scheduling	C.M. Meyer; A.L. Kok; H. Kopfer; J.M.J.
		from a distributed decision making perspective	Schutten
	<b>_</b>		
288	2010	Lead time anticipation in Supply Chain Operations Planning	Michiel Jansen; Ton G. de Kok; Jan C.
			Fransoo
287	2009	Inventory Models with Lateral Transshipments: A Review	Colin Paterson; Gudrun Kiesmuller;
			Ruud Teunter; Kevin Glazebrook
286	2009	Efficiency evaluation for pooling resources in health care	P.T. Vanberkel; R.J. Boucherie; E.W.
	<b>_</b>		Hans; J.L. Hurink; N. Litvak
285	2009	A Survey of Health Care Models that Encompass Multiple	P.T. Vanberkel; R.J. Boucherie; E.W.
		Departments	Hans; J.L. Hurink; N. Litvak
284	2009	Supporting Process Control in Business Collaborations	S. Angelov; K. Vidyasankar; J. Vonk; P.
202	2000		Greten
283	2009	Inventory Control with Partial Batch Ordering	O. Alp; W.I. Hun; I. Tan
282	2009	Translating Safe Petri Nets to Statecharts in a Structure-	R. Eshuis
201	2000	Preserving way	
281	2009	The link between product data model and process model	J.J.C.L. Vogelaar; H.A. Reijers
200	2000	Inventory planning for spare parts patworks with delivery	LC Deiinen T. Ten C. Lyon Heutum
280	2009	time requirements	I.C. Reijnen; T. Tan; G.J. Van Houtum
270	2000	Con Evolution of Domand and Supply under Competition	R Vermeulen: A.C. de Kek
279	2009	co-evolution of Demand and Supply under competition	B. Vermeulen, A.G. de Kok
278	2010	Toward Meso-level Product-Market Network Indices for	B Vermeulen A.G. de Kok
270	2010	Strategic Product Selection and (Re)Design Guidelines over	b. Vermeuen, A.G. de Kok
		the Product Life-Cycle	
277	2009	An Efficient Method to Construct Minimal Protocol Adaptors	R Seguel R Eshuis P Grefen
277	2005	An Emelent Wiethou to construct Minimar Protocol Adaptors	
276	2009	Coordinating Supply Chains: a Bilevel Programming	Ton G. de Kok. Gabriella Muratore
270	2005	Approach	
275	2009	Inventory redistribution for fashion products under demand	G P Kiesmuller S Minner
275	2005	narameter undate	
274	2009	Comparing Markov chains: Combining aggregation and	A Busic I M H Vliegen A Scheller-Wolf
274	2005	nrecedence relations annlied to sets of states	A busic, invint. Vilegen, A. Schener Woh
273	2009	Separate tools or tool kits: an exploratory study of	I.M.H. Vliegen, P.A.M. Kleingeld, G.I.
		engineers' preferences	van Houtum
272	2009	An Exact Solution Procedure for Multi-Item Two-Echelon	
_		Spare Parts Inventory Control Problem with Batch Ordering	

Nr.	Year	Title	Author(s)
271	2009	Distributed Decision Making in Combined Vehicle Routing	C.M. Meyer, H. Kopfer, A.L. Kok, M.
		and Break Scheduling	Schutten
270	2009	Dynamic Programming Algorithm for the Vehicle Routing	A.L. Kok, C.M. Meyer, H. Kopfer, J.M.J.
		Problem with Time Windows and EC Social Legislation	Schutten
269	2009	Similarity of Business Process Models: Metics and Evaluation	Remco Dijkman, Marlon Dumas,
			Boudewijn van Dongen, Reina Kaarik,
			Jan Mendling
267	2009	Vehicle routing under time-dependent travel times: the	A.L. Kok, E.W. Hans, J.M.J. Schutten
		impact of congestion avoidance	
266	2009	Restricted dynamic programming: a flexible framework for	J. Gromicho; J.J. van Hoorn; A.L. Kok;
		solving realistic VRPs	J.M.J. Schutten;

Working Papers published before 2009 see: http://beta.ieis.tue.nl